ASSET PURCHASE AGREEMENT

BETWEEN

SHARPER IMAGE CORPORATION

AND

A JOINT VENTURE OF HILCO MERCHANT RESOURCES, LLC HILCO CONSUMER CAPITAL, LLC GORDON BROTHERS RETAIL PARTNERS, LLC AND GB BRANDS, LLC

Dated as of May 13, 2008

TABLE OF CONTENTS

Page

Article I	DEFINITIONS	2
1.1	Certain Definitions	2
1.2	Terms Defined Elsewhere in this Agreement	5
1.3	Other Definitional and Interpretive Matters	6
Article II	PURCHASE AND SALE OF ASSETS; ASSUMPTION OF LIABILITIES	7
2.1	Purchase and Sale of Assets	7
2.2	Agency Agreement	7
2.3	Excluded Assets	
2.4	Assumption of Liabilities	8
2.5	Excluded Liabilities	9
2.6	Cure Amounts	9
2.7	Rejection of Certain Executory Contracts	9
2.8	Further Conveyances and Assumptions	9
2.9	Bulk Sales Laws	
Article III	CONSIDERATION	
3.1	Consideration	
3.2	Payment of Purchase Price	
Article IV	CLOSING AND TERMINATION	
4.1	Closing Date	
4.2	Deliveries by Seller	
4.3	Deliveries by Purchaser	11
4.4	Termination of Agreement	11
4.5	Procedure Upon Termination	12
4.6	Effect of Termination	12
Article V	REPRESENTATIONS AND WARRANTIES OF SELLER	12
5.1	Authorization of Agreement	12
5.2	Conflicts; Consents of Third Parties	13
5.3	Assets Sold Free and Clear	13
5.4	Purchased Intellectual Property	14
5.5	Financial Advisors	15

TABLE OF CONTENTS (continued)

Page

	5.6	No Other Representations or Warranties; Schedules	15
Article	VI	REPRESENTATIONS AND WARRANTIES OF PURCHASER	15
	6.1	Organization and Good Standing	15
	6.2	Authorization of Agreement	15
	6.3	Conflicts; Consents of Third Parties	16
	6.4	Litigation	16
	6.5	Financial Advisors	16
	6.6	Financial Capability	16
	6.7	Condition of the Business	17
Article	VII	BANKRUPTCY COURT MATTERS	17
	7.1	Competing Transaction	17
	7.2	Bankruptcy Court Filings	18
Article	VIII	COVENANTS	18
	8.1	Access to Information	18
	8.2	Conduct of the Business Pending the Closing	18
	8.3	Consents	19
	8.4	Regulatory Approvals.	19
	8.5	Further Assurances	19
	8.6	Confidentiality	20
	8.7	Customer Lists	20
	8.8	Preservation of Records	20
	8.9	Publicity	20
	8.10	Use of Name	20
	8.11	Supplementation and Amendment of Schedules	21
	8.12	Court Order	21
	8.13	Adequate Assurance of Future Performance	21
	8.14	Purchaser Covenants after Closing	21
Article	IX	EMPLOYEES AND EMPLOYEE BENEFITS	22
	9.1	Employment	22
Article	Х	CONDITIONS TO CLOSING	22
	10.1	Conditions Precedent to Obligations of Purchaser	22

TABLE OF CONTENTS (continued)

Page

10.2	Conditions Precedent to Obligations of Seller	23
10.3	Conditions Precedent to Obligations of Purchaser and Seller	
10.4	Frustration of Closing Conditions	
Article XI	NO SURVIVAL	
11.1	No Survival of Representations and Warranties	
11.2	No Consequential Damages	
Article XII	TAXES	
12.1	Transfer Taxes	
12.2	Prorations	
Article XIII	MISCELLANEOUS	
13.1	Expenses	
13.2	Injunctive Relief	
13.3	Submission to Jurisdiction; Consent to Service of Process	
13.4	Waiver of Right to Trial by Jury	
13.5	Entire Agreement; Amendments and Waivers	
13.6	Governing Law	
13.7	Notices	
13.8	Severability	
13.9	Binding Effect; Assignment	
13.10	Conflict of Terms	
13.11	Non-Recourse.	
13.12	Joint and Several Liability of the Purchaser	
13.13	Counterparts	
Schedules		

1.1(a)	Purchased Contracts
2.6(a)	Cure Amounts
2.6(b)	Core IP Contracts
2.7	Executory Contract and Leases Designated for Rejection
5.4(a)	Purchased Intellectual Property
5.4(b)	Domain Names
5.4(c)	IP Contracts
5.4(e)	Restrictions on Purchased Intellectual Property
5.5	Financial Advisors

TABLE OF CONTENTS (continued)

Page

Exhibits

А	Agency Agreement

B Form of Assumption and Assignment Agreement

ASSET PURCHASE AGREEMENT

ASSET PURCHASE AGREEMENT, dated as of May 13, 2008 (this "<u>Agreement</u>"), between Sharper Image Corporation, a Delaware corporation (together with any affiliates, "<u>Seller</u>"), and a joint venture formed among: (a) Gordon Brothers Retail Partners, (b) GB Brands, LLC, (c) Hilco Merchant Resources, LLC, and (d) Hilco Consumer Capital, LLC (collectively, the "<u>Purchaser</u>").

WITNESSETH:

WHEREAS, Seller is a debtor in possession under title 11 of the United States Code, 11 U.S.C. § 101 et seq. (the "<u>Bankruptcy Code</u>"), and filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code on February 19, 2008, in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>") (Case No. 08-10322(KG)) (the "<u>Bankruptcy Case</u>");

WHEREAS, Seller presently conducts the Business;

WHEREAS, Seller desires to sell, transfer and assign to Purchaser, and Purchaser desires to purchase, acquire and assume from Seller, pursuant to Sections 105, 363 and 365 of the Bankruptcy Code, all of the Purchased Assets and Assumed Liabilities, all as more specifically provided herein;

WHEREAS, on or about April 25, 2008, Seller filed the Sale Motion in the Bankruptcy Case requesting, <u>inter alia</u>, authorization to (i) enter into this Agreement and (ii) sell certain of its remaining assets, subject to Seller's solicitation and receipt of higher and/or better offers consistent with Section 7 hereof; and

WHEREAS, contemporaneously herewith, Seller and Purchaser have entered into that certain Agency Agreement (the "<u>Agency Agreement</u>"), dated of even date, substantially in the form attached hereto as **Exhibit A**; and

WHEREAS, certain terms used in this Agreement are defined in Section 1.1.

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements hereinafter contained, the parties hereby agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Certain Definitions</u>.

For purposes of this Agreement, the following terms shall have the meanings specified in this <u>Section 1.1</u>:

"<u>Advertising and Marketing Materials</u>" means, to the extent owned by Seller as of the date hereof, any and all of Seller's advertising materials and related designs, patterns, drawings and specifications, pricing and cost documentation, and marketing materials, including historical or archival materials held by Seller in inventory.

"<u>Affiliate</u>" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, such Person, and the term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by contract or otherwise.

"<u>Business</u>" means the business of Seller, including, without limitation, the sale of specialty retail products through retail stores, via catalog, and over the internet.

"<u>Business Day</u>" means any day of the year on which national banking institutions in New York are open to the public for conducting business and are not required or authorized to close.

"<u>Contract</u>" means any written contract, indenture, note, bond, lease or other agreement.

"<u>Core IP Contract</u>" means the IP Contracts set forth on <u>Schedule 2.6(a)</u> hereto; <u>provided</u>, <u>however</u>, that Purchaser shall have the right to remove any of such IP Contracts from Schedule 2.6(a) prior to the Sale Hearing, which removal shall not result in a reduction in the Purchase Price.

"<u>Deposit</u>" means a cash deposit equal to \$2.5 million to be delivered by the Purchaser to the Seller prior to the hearing before the Bankruptcy Court to approve the procedures relating to the Sale Motion, including, without limitation, approval of the Breakup Fee and Expense Reimbursement.

"<u>Documents</u>" means all files, documents, instruments, papers, books, reports, records, tapes, microfilms, photographs, letters, budgets, forecasts, ledgers, journals, title policies, customer lists, regulatory filings, operating data and plans, technical documentation (design specifications, functional requirements, operating instructions, logic manuals, flow charts, etc.), user documentation (installation guides, user manuals, training materials, release notes, working papers, etc.), marketing documentation (sales brochures, flyers, pamphlets, web

pages, etc.), and other similar materials related to the Business and the Purchased Assets in each case whether or not in electronic form.

"<u>Employees</u>" means all individuals, as of the date hereof, whether or not actively at work as of the date hereof, who are employed by Seller in connection with the Business, together with individuals who are hired in respect of the Business after the date hereof and prior to the Closing.

"<u>Excluded Contracts</u>" means any Contract to which the Seller is a party with the exception of the Purchased Contracts.

"<u>Governmental Body</u>" means any government or governmental or regulatory body thereof, or political subdivision thereof, whether foreign, federal, state, or local, or any agency, instrumentality or authority thereof, or any court or arbitrator (public or private).

"<u>IP Contracts</u>" means all written agreements containing a grant, either from a third party to the Seller or from the Seller to any third party, of the right to use any Purchased Intellectual Property, Technology or Software.

"<u>Law</u>" means any federal, state, local or foreign law, statute, code, ordinance, rule or regulation.

"<u>Legal Proceeding</u>" means any judicial, administrative or arbitral actions, suits, proceedings (public or private) or claims or any proceedings by or before a Governmental Body.

"<u>Liability</u>" means any debt, liability or obligation (whether direct or indirect, known or unknown, absolute or contingent, accrued or unaccrued, liquidated or unliquidated, or due or to become due), and including all costs and expenses relating thereto.

"<u>Lien</u>" means any lien, encumbrance, pledge, mortgage, deed of trust, security interest, claim, lease, charge, option, right of first refusal, easement, servitude, proxy, voting trust or agreement, transfer restriction under any shareholder or similar agreement or encumbrance.

"<u>Material Adverse Effect</u>" means (i) a material adverse effect on the business, assets, properties, results of operations or financial condition of Seller, or (ii) a material adverse effect on the ability of Seller to consummate the transactions contemplated by this Agreement or perform its obligations under this Agreement, other than an effect resulting from an Excluded Matter. "<u>Excluded Matter</u>" means any one or more of the following: (i) the effect of any change in the United States or foreign economies or securities or financial markets in general; (ii) the effect of any change that generally affects any industry in which Seller operates; (iii) the effect of any change arising in connection with earthquakes, hostilities, acts of war, sabotage or terrorism or military actions or any escalation or material worsening of any such hostilities, acts of war, sabotage or terrorism or military action taken by Purchaser or its Affiliates with respect to the transactions contemplated hereby or with respect to Seller, including its employees; (v) any matter of which Purchaser is aware on the date hereof; (vi) the effect of any changes in applicable Laws or accounting rules; or (vii) any effect resulting from the public announcement of this Agreement,

compliance with terms of this Agreement or the consummation of the transactions contemplated by this Agreement; or (viii) any effect resulting from the filing of the Bankruptcy Case and reasonably anticipated effects thereof.

"<u>Order</u>" means any order, injunction, judgment, decree, ruling, writ, assessment or arbitration award of a Governmental Body.

"<u>Ordinary Course of Business</u>" means the ordinary and usual course of normal day-to-day operations of the Business through the date hereof consistent with past practice.

"<u>Permits</u>" means any approvals, authorizations, consents, licenses, permits or certificates of a Governmental Body.

"<u>Person</u>" means any individual, corporation, limited liability company, partnership, firm, joint venture, association, joint-stock company, trust, unincorporated organization, Governmental Body or other entity.

"<u>Products</u>" means any and all products developed, manufactured, marketed or sold by Seller, whether work in progress or in final form.

"<u>Procedures Order</u>" means an order of the Bankruptcy Court, as requested by the Seller, and reasonably acceptable to Purchaser, that, among other things, establishes a date by which Competing Offers must be submitted by bidders and establishes procedures for the auction process.

"<u>Purchased Contracts</u>" means the Contracts set forth on <u>Schedule 1.1(a)</u>, to be delivered by the Purchaser to the Seller no later than three (3) business days prior to the auction.

"Purchased Intellectual Property" means all of the Seller's right, title and interest in and to any intellectual property and intellectual property rights, including, without limitation: (i) all patents, patent applications and patent disclosures, including continuations, divisionals, continuations-in-part, or reissues of patent applications and patents issuing thereon (collectively, "Patents"), (ii) all trademarks, service marks, trade names, service names, brand names, all trade dress rights, logos, Internet domain names and corporate names and general intangibles of a like nature, together with the goodwill associated with any of the foregoing, and all applications, registrations and renewals thereof (collectively, "Marks"), (iii) copyrights and registrations and applications therefor and works of authorship, and mask work rights, in each case used primarily in connection with the Business, (collectively, "Copyrights"), (iv) all Software and Technology and Advertising and Marketing Materials of Seller, (v) inventions, whether patentable or unpatentable and whether or not reduced to practice (to the extent any protectable right exists in any such invention not reduced to practice), and all improvements in existence at the Closing relating to the inventions, (vi) trade secrets and confidential business information, including, without limitation, ideas, research and development, know-how, formulas, compositions, manufacturing and production processes and techniques, technical data, databases, customer lists and contact information, supplier lists, and business and marketing plans and proposals; and (vii) other proprietary rights and copies and tangible expressions thereof.

"<u>Sale Motion</u>" means the motion or supplements thereto of Seller seeking approval from the Bankruptcy Court for entry of the Procedures Order and Sale Order.

"<u>Sale Order</u>" shall be an order or orders of the Bankruptcy Court in form and substance reasonably acceptable to Purchaser and Seller approving this Agreement and all of the respective terms and conditions hereof, and approving and authorizing Seller to consummate the transactions contemplated hereby.

"<u>Software</u>" means, except to the extent generally available for purchase from a third Person, any and all (i) computer programs, including any and all software implementations of algorithms, models and methodologies, whether in source code or object code, (ii) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (iii) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons, and (iv) all documentation including user manuals and other training documentation related to any of the foregoing.

"<u>Subsidiary</u>" means any Person of which a majority of the outstanding voting securities or other voting equity interests are owned, directly or indirectly, by Purchaser.

"<u>Taxes</u>" means (i) all federal, state, local or foreign taxes, charges or other assessments, including, without limitation, all net income, gross receipts, capital, sales, use, ad valorem, value added, transfer, franchise, profits, inventory, capital stock, license, withholding, payroll, employment, social security, unemployment, excise, severance, stamp, occupation, property and estimated taxes, and (ii) all interest, penalties, fines, additions to tax or additional amounts imposed by any taxing authority in connection with any item described in clause (i).

"<u>Tax Return</u>" means all returns, declarations, reports, estimates, information returns and statements required to be filed in respect of any Taxes.

"<u>Technology</u>" means, collectively, all designs, formulae, algorithms, procedures, methods, techniques, ideas, know-how, research and development, technical data, programs, subroutines, tools, materials, specifications, processes, inventions (whether patentable or unpatentable and whether or not reduced to practice), apparatus, creations, improvements, works of authorship and other similar materials, and all recordings, graphs, drawings, reports, analyses, and other writings, and other tangible embodiments of the foregoing, in any form whether or not specifically listed herein, and all related technology, that are used in, incorporated in, embodied in, displayed by or relate to, or are used or useful in the design, development, reproduction, maintenance or modification of, any of the Products. 1.2 <u>Terms Defined Elsewhere in this Agreement</u>. For purposes of this Agreement, the following terms have meanings set forth in the sections indicated:

Term	Section
Agency Agreement	Recitals
Assumed Liabilities	3.1
Bankruptcy Case	Recitals
Bankruptcy Code	Recitals
Bankruptcy Court	Recitals
Breakup Fee	7.1(b)
Closing	4.1
Closing Date	4.1
Competing Offer	7.1
Confidential Information	8.6
Confidentiality Agreement	8.6
Expense Reimbursement Excluded Assets	7.1(b) 2.2 2.1
Purchased Assets Purchase Price Purchaser	3.1 Recitals
Purchaser Documents	6.2
Seller	Recitals
Seller Documents	5.1
Seller Marks	8.10
Termination Date	4.4(a)
Transfer Taxes	12.1

1.3 Other Definitional and Interpretive Matters

(a) Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

<u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day.

Dollars. Any reference in this Agreement to \$ shall mean U.S. dollars.

<u>Exhibits/Schedules</u>. All Exhibits and Schedules annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any matter or item disclosed on one schedule shall be deemed to have been disclosed on each other schedule. Any capitalized terms used in any Schedule or Exhibit but not otherwise defined therein shall be defined as set forth in this Agreement.

<u>Gender and Number</u>. Any reference in this Agreement to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.

<u>Headings</u>. The provision of a Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "Section" are to the corresponding Section of this Agreement unless otherwise specified.

<u>Herein</u>. The words such as "<u>herein</u>," "<u>hereinafter</u>," "<u>hereof</u>," and "<u>hereunder</u>" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.

<u>Including</u>. The word "<u>including</u>" or any variation thereof means "<u>including</u>, <u>without limitation</u>" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

(b) The parties hereto have participated jointly in the negotiation and drafting of this Agreement and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the parties hereto and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provision of this Agreement.

ARTICLE II

PURCHASE AND SALE OF ASSETS; ASSUMPTION OF LIABILITIES

2.1 <u>Purchase and Sale of Assets</u>. On the terms and subject to the conditions set forth in this Agreement, at the Closing, Purchaser shall purchase, acquire and accept from Seller, and Seller shall sell, transfer, assign, convey and deliver to Purchaser, all of Seller's right, title and interest in, to and under the Purchased Assets. "<u>Purchased Assets</u>" shall mean the following assets of Seller (but excluding Excluded Assets) as of the Closing to the extent related to the Business:

(a) the right, at Purchaser's discretion, to sell or dispose, by means of a store closing sale (the "<u>Liquidation</u>") of Seller's inventory (the "<u>Merchandise</u>"), together with all furniture, fixtures and equipment ("<u>FFE</u>") and supplies, in connection with the Business, on the terms and conditions set forth in the Agency Agreement;

- (b) the Purchased Intellectual Property;
- (c) the Purchased Contracts
- (f) any and all assets relating to the Seller's catalog or e-commerce business;

and

(f) the Advertising and Marketing Materials

2.2 <u>Agency Agreement</u>. Notwithstanding anything to the contrary set forth herein, all terms and conditions relating to the Merchandise, the FFE, and the Liquidation shall be governed by the Agency Agreement. In the event of any conflict between this Agreement and the Agency Agreement relating thereto, the Agency Agreement shall govern.

2.3 <u>Excluded Assets</u>. Nothing herein contained shall be deemed to sell, transfer, assign or convey the Excluded Assets to Purchaser, and Seller shall retain all right, title and interest to, in and under the Excluded Assets. "<u>Excluded Assets</u>" shall mean all assets, properties, interests and rights of Seller other than the Purchased Assets, including, without limitation, each of the following assets:

(a) all cash, cash equivalents, bank deposits or similar cash items of Seller;

(b) all of Seller's deposits or prepaid charges and expenses paid in connection with or relating to any Excluded Assets;

(c) the Excluded Contracts, including any accounts receivable arising out of or in connection with any Excluded Contract;

(d) any (i) books and records that Seller is required by Law to retain or that Seller determines are necessary or advisable to retain including, without limitation, Tax Returns, financial statements, and corporate or other entity filings; <u>provided</u>, <u>however</u>, that Purchaser shall have the right to make copies of any portions of such retained books and records that relate to the Business or any of the Purchased Assets; (ii) information management systems of Seller, other than those used or held for use in the conduct of the Business; and (iii) documents relating to proposals to acquire the Business by Persons other than Purchaser.

(e) any claim, right or interest of Seller in or to any refund, rebate, abatement or other recovery for Taxes, together with any interest due thereon or penalty rebate arising therefrom, for any Tax period (or portion thereof) ending on or before the Closing Date;

(f) all insurance policies or rights to proceeds thereof relating to the assets, properties, business or operations of Seller; and

(g) any rights, claims or causes of action of Seller against third parties relating to the assets, properties, business or operations of Seller arising out of events occurring on or prior to the Closing Date, including, but not limited to, commercial tort causes of action and all causes of action under chapter 5 of the Bankruptcy Code.

2.4 <u>Assumption of Liabilities</u>. On the terms and subject to the conditions set forth in this Agreement and the Agency Agreement, at the Closing, Purchaser shall assume, effective as of the Closing Date, the following Liabilities (the "<u>Assumed Liabilities</u>"):

(a) all Liabilities of Seller under the Purchased Contracts;

(b) all Liabilities arising from or relating to the Liquidation, on the terms and conditions set forth in the Agency Agreement;

(c) all Transfer Taxes, if any, but only to the extent not exempt under the Bankruptcy Code, as applicable to the transfer of the Purchased Assets pursuant to this Agreement and, further, solely to the extent applicable law may impose any Transfer Tax liability on the purchaser of such assets;

(d) all Liabilities relating to amounts required to be paid by Purchaser hereunder; and

Closing.

(e) all other Liabilities with respect to the Purchased Assets arising after the

2.5 <u>Excluded Liabilities</u>. Except as specified in foregoing Section 2.4, Purchaser does not assume any other Liabilities or obligations of Seller.

2.6 Cure Amounts. At Closing and pursuant to Section 365 of the Bankruptcy Code, Seller shall assume and assign to Purchaser and Purchaser shall assume from Seller, the Purchased Contracts. Except as otherwise set forth in any Bankruptcy Court order approving this Agreement and authorizing the sale of the Purchased Assets, the cure amounts, as determined by the Bankruptcy Court, if any, necessary to cure all defaults, if any, (including, if any, and to the extent allowed by the Bankruptcy Court, all actual or pecuniary losses that have resulted from such defaults under the Purchased Contracts), shall be paid by Purchaser, on or before Closing, and not by Seller and Seller shall have no liability therefore; provided, however, that Purchaser shall not be obligated to cure any such default that is (a) (i) greater than 110% of the total amount set forth in the cure amounts schedule attached as Schedule 2.6(a) hereof or (ii) in excess of the total amount set forth in the cure amount schedule that is to be dated May 21, 2008, whichever is greater (with such excess being Seller's sole responsibility), or (b) that is in dispute as of the Closing Date if an escrow or other commercially reasonable arrangement with respect thereto shall have been established at or following the Closing pursuant to an order of the Bankruptcy Court. Purchaser shall have the right to a reduction of the Purchase Price (with an amount agreed to by Seller and Purchaser or as determined by the Bankruptcy Court) as a result of the failure by Seller to assume and assign to Purchaser any Core IP Contract.

2.7 <u>Rejection of Certain Executory Contracts</u>. As part of the Sale Order, the Bankruptcy Court shall authorize the Seller's rejection of the executory contracts and/or leases (other than non-residential real property leases) listed on <u>Schedule 2.7</u> hereof.

2.8 <u>Further Conveyances and Assumptions</u>.

(a) From time to time following the Closing, Seller and Purchaser shall, and shall cause their respective Affiliates to, execute, acknowledge and deliver all such further conveyances, notices, assumptions, releases and acquaintances and such other instruments, and shall take such further actions, as may be reasonably necessary or appropriate to assure fully to Purchaser and its respective successors or assigns, all of the properties, rights, titles, interests, estates, remedies, powers and privileges intended to be conveyed to Purchaser under this Agreement and the Seller Documents and to assure fully to Seller and its Affiliates and their successors and assigns, the assumption of the liabilities and obligations intended to be assumed

by Purchaser under this Agreement and the Seller Documents, and to otherwise make effective the transactions contemplated hereby and thereby.

2.9 <u>Bulk Sales Laws</u>. Purchaser hereby waives compliance by Seller with the requirements and provisions of any "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the sale and transfer of any or all of the Purchased Assets to Purchaser.

ARTICLE III

CONSIDERATION

3.1 <u>Consideration</u>. The aggregate consideration for the Purchased Assets shall be (a) an amount in cash equal to \$51,250,000, less all payments received by Seller on or after May 1, 2008 under the Core IP Contracts, and (b) the assumption of the Assumed Liabilities (collectively, the "<u>Purchase Price</u>").

3.2 <u>Payment of Purchase Price</u>. On the Closing Date, Purchaser shall pay the Purchase Price minus the Deposit to Seller, which shall be paid by wire transfer of immediately available funds into an account designated by Seller, subject to the terms of the Agency Agreement.

ARTICLE IV

CLOSING AND TERMINATION

4.1 <u>Closing Date</u>. Subject to the satisfaction of the conditions set forth in <u>Sections</u> <u>10.1, 10.2</u> and <u>10.3</u> hereof (or the waiver thereof by the party entitled to waive that condition), the sale of the Purchased Assets and the assumption of the Assumed Liabilities provided for in <u>Article II</u> hereof (the "<u>Closing</u>") shall take place at the offices of Weil, Gotshal & Manges LLP located at 767 Fifth Avenue, New York, New York (or at such other place as the parties may designate in writing) at 12 p.m. (eastern time) on the date that is two (2) Business Days following the satisfaction or waiver of the conditions set forth in <u>Article X</u> (other than conditions that by their nature are to be satisfied at the Closing, but subject to the satisfaction or waiver of such conditions), unless another time or date, or both, are agreed to in writing by the parties hereto. The date on which the Closing shall be held is referred to in this Agreement as the "<u>Closing Date</u>." Unless otherwise agreed by the parties in writing, the Closing shall be deemed effective and all right, title and interest of Seller to be acquired by Purchaser hereunder shall be considered to have passed to Purchaser as of 12:01 a.m. (eastern time) on the Closing Date.

4.2 <u>Deliveries by Seller</u>. At the Closing, Seller shall deliver to Purchaser:

(a) duly executed assumption and assignment agreements in the form of <u>Exhibit B</u> hereto and duly executed assignments of the U.S. trademark registrations and applications included in the Purchased Intellectual Property, in a form suitable for recording in the U.S. Trademark office, and general assignments of other Purchased Intellectual Property; and

(b) all other instruments of conveyance and transfer, in form and substance reasonably acceptable to Purchaser, as may be necessary to convey the Purchased Assets to Purchaser.

4.3 <u>Deliveries by Purchaser</u>. At the Closing, Purchaser shall deliver to Seller:

(a) the Purchase Price in immediately available funds as set forth in Section3.2 hereof;

(b) duly executed counterparts to the Agreement, the Agency Agreement and the assignment and assumption agreements referred to in Section 4.2(a); and

(c) such other documents, instruments and certificates as Seller may reasonably request.

4.4 <u>Termination of Agreement</u>. This Agreement may be terminated prior to the Closing as follows:

(a) by Purchaser or Seller, if the Closing shall not have occurred by the close of business on June 6, 2008 (the "<u>Termination Date</u>"); <u>provided</u>, <u>however</u>, that if the Closing shall not have occurred on or before the Termination Date due to a material breach of any representations, warranties, covenants or agreements contained in this Agreement by Purchaser or Seller, then the breaching party may not terminate this Agreement pursuant to this <u>Section 4.4(a)</u>;

(b) by mutual written consent of Seller and Purchaser;

(c) by Purchaser, if any of the conditions to the obligations of Purchaser set forth in <u>Sections 10.1</u> and <u>10.3</u> shall have become incapable of fulfillment other than as a result of a breach by Purchaser of any covenant or agreement contained in this Agreement, and such condition is not waived by Purchaser;

(d) by Seller, if any condition to the obligations of Seller set forth in Sections 10.2 and 10.3 shall have become incapable of fulfillment other than as a result of a breach by Seller of any covenant or agreement contained in this Agreement, and such condition is not waived by Seller;

(e) by Purchaser, if there shall be a breach by Seller of any representation or warranty, or any covenant or agreement contained in this Agreement which would result in a failure of a condition set forth in Section 10.1 or 10.3 and which breach cannot be cured or has not been cured by the earlier of (i) twenty (20) Business Days after the giving of written notice by Purchaser to Seller of such breach and (ii) the Termination Date;

(f) by Seller, if there shall be a breach by Purchaser of any representation or warranty, or any covenant or agreement contained in this Agreement which would result in a failure of a condition set forth in Section 10.2 or 10.3 and which breach cannot be cured or has

not been cured by the earlier of (i) twenty (20) Business Days after the giving of written notice by Seller to Purchaser of such breach and (ii) the Termination Date;

(g) by Seller or Purchaser if there shall be in effect a final nonappealable Order of a Governmental Body of competent jurisdiction restraining, enjoining or otherwise prohibiting the consummation of the transactions contemplated hereby; it being agreed that the parties hereto shall promptly appeal any adverse determination which is not nonappealable (and pursue such appeal with reasonable diligence); or

(h) by Purchaser or Seller, if the Bankruptcy Court shall enter an order approving a Competing Offer, subject to the limitations set forth in the Procedures Order.

4.5 <u>Procedure Upon Termination</u>. In the event of termination and abandonment by Purchaser or Seller, or both, pursuant to <u>Section 4.4</u> hereof, written notice thereof shall forthwith be given to the other party or parties, and this Agreement shall terminate, and the purchase of the Purchased Assets hereunder shall be abandoned, without further action by Purchaser or Seller. If this Agreement is terminated as provided herein each party shall redeliver or destroy all confidential non-public documents, work papers and other material of any other party relating to the transactions contemplated hereby, whether so obtained before or after the execution hereof, to the party furnishing the same.

4.6 <u>Effect of Termination</u>.

(a) In the event that this Agreement is validly terminated as provided herein, then each of the parties shall be relieved of its duties and obligations arising under this Agreement after the date of such termination and such termination shall be without liability to Purchaser or Seller; <u>provided</u>, <u>however</u>, that the obligations of the parties set forth in Section 7.1(b) and <u>Article XIII</u> hereof shall survive any such termination and shall be enforceable hereunder.

(b) Nothing in this <u>Section 4.6</u> shall relieve Purchaser or Seller of any liability for a breach of this Agreement prior to the date of termination. The damages recoverable by the non-breaching party shall include all attorneys' fees reasonably incurred by such party in connection with the transactions contemplated hereby.

(c) The Confidentiality Agreement shall survive any termination of this Agreement and nothing in this <u>Section 4.6</u> shall relieve Purchaser or Seller of their obligations under the Confidentiality Agreement.

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF SELLER

Seller hereby represents and warrants to Purchaser that:

5.1 <u>Authorization of Agreement</u>. Except for such authorization as is required by the Bankruptcy Court (as hereinafter provided for), Seller has all requisite power, authority and legal

capacity to execute and deliver this Agreement and Seller has all requisite power, authority and legal capacity to execute and deliver each other agreement, document, or instrument or certificate contemplated by this Agreement or to be executed by Seller in connection with the consummation of the transactions contemplated by this Agreement (the "Seller Documents"), to perform its respective obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. The execution and delivery of this Agreement and the Seller Documents and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all requisite corporate action on the part of Seller. This Agreement has been, and each of the Seller Documents will be at or prior to the Closing, duly and validly executed and delivered by Seller and (assuming the due authorization, execution and delivery by the other parties hereto and thereto, the entry of the Sale Order) this Agreement constitutes, and each of the Seller Documents when so executed and delivered will constitute, legal, valid and binding obligations of Seller enforceable against Seller in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

5.2 <u>Conflicts; Consents of Third Parties</u>.

(a) To the knowledge of Seller, none of the execution and delivery by Seller of this Agreement or by Seller of the Seller Documents, the consummation of the transactions contemplated hereby or thereby, or compliance by Seller with any of the provisions hereof or thereof will conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination or cancellation under any provision of (i) the certificate of incorporation and by-laws or comparable organizational documents of Seller; (ii) subject to entry of the Sale Order, any Contract or Permit to which Seller is a party or by which any of the properties or assets of Seller are bound; (iii) subject to entry of the Sale Order, any Order of any Governmental Body applicable to Seller or any of the properties or assets of Seller as of the date hereof; or (iv) subject to entry of the Sale Order, any applicable Law, other than, in the case of clauses (ii), (iii) and (iv), such conflicts, violations, defaults, terminations or cancellations that would not have a Material Adverse Effect.

(b) To the knowledge of Seller, no consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Body is required on the part of Seller in connection with the execution and delivery of this Agreement or the Seller Documents, the compliance by Seller with any of the provisions hereof or thereof, the consummation of the transactions contemplated hereby or thereby or the taking by Seller of any other action contemplated hereby or thereby, except for (i) the entry of the Sale Order, and (ii) such other consents, waivers, approvals, Orders, Permits, authorizations, declarations, filings and notifications, the failure of which to obtain or make would not have a Material Adverse Effect.

5.3 <u>Assets Sold Free and Clear</u>. Upon the execution and delivery by Seller of this Agreement and the Seller Documents and the consummation of the transactions contemplated

hereby or thereby, following Bankruptcy Court approval thereof, Purchaser will receive all of Seller's right, title and interest to the Purchased Assets free and clear of any and all Liens pursuant to Sections 363 and 365 of the Bankruptcy Code.

5.4 <u>Purchased Intellectual Property</u>

(a) <u>Schedule 5.4(a)</u> sets forth a true, correct and complete list of all Purchased Intellectual Property that is registered or for which an application for registration is pending (together with the country, title and registration or application number as applicable) and all material unregistered Marks. Except as disclosed on <u>Schedule 5.4(a)</u>, to the knowledge of Seller, except as would not have a Material Adverse Effect, (i) Seller owns all right, title and interest in and to the Purchased Intellectual Property, (ii) the registered Purchased Intellectual Property is subsisting, valid, enforceable and all renewal or maintenance fees and related filings have been duly made (except where Seller in good faith decided to abandon non-material registered Purchased Intellectual Property), and (iii) the Purchased Intellectual Property constitutes all of the intellectual property rights necessary for Seller to operate the Business as it is presently conducted.

(b) <u>Schedule 5.4(b)</u> sets forth a true, correct and complete list of domain names owned or controlled by Seller.

(c) <u>Schedule 5.4(c)</u> sets forth a true, correct and complete list of all material IP Contracts, except for licenses of Software (i) generally available for an annual or one-time license fee of no more than \$10,000 in the aggregate, (ii) distributed as "freeware" or (iii) distributed via Internet access without charge. To the knowledge of Seller, Seller and any applicable third party are in material compliance with the terms of all such IP Contracts. <u>Schedule 5.4(c)</u> also sets forth true, accurate and complete statements of royalties, license fees and other payments received by Seller, and the date of receipt, relating to the Core IP Contracts, including copies of all sales reports received by Seller under the Core IP Contracts. To the knowledge of Seller, except as disclosed on Schedule 5.4(c), all third party Software used in the Business is the subject of a valid license agreement.

(d) To the knowledge of Seller, (i) no Person is infringing or otherwise violating any material Patent(s), Mark(s), Copyright(s), trade secret(s) or other proprietary right(s) of Seller; (ii) no proceeding or claims have been asserted by Seller alleging that any person or entity is infringing upon or otherwise violating any Patent(s), Mark(s), Copyright(s), trade secret(s) or other proprietary right(s) of Seller; (iii) to the knowledge of Seller, there has been no unauthorized use or disclosure or any material trade secret of Seller by any current or former employee, licensee, distributor or consultant of Seller or by any other third party; and (iv) no claims are pending or, to the knowledge of Seller, threatened, against Seller based on any claim or allegation that the operation of the Business infringes, misappropriates, dilutes, or otherwise violates any intellectual property, privacy or publicity right of any third party.

(e) Except as set forth on <u>Schedule 5.4(e)</u>, there are no agreements (other than licenses of the Purchased Intellectual Property), orders or judgments which would restrict or adversely affect Purchaser's right to use, license or exploit the Purchased Intellectual Property.

5.5 <u>Financial Advisors</u>. Except as set forth on <u>Schedule 5.5</u>, no Person has acted, directly or indirectly, as a broker, finder or financial advisor for Seller or any of its Subsidiaries in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment from Purchaser in respect thereof.

No Other Representations or Warranties; Schedules. Except for the 5.6 representations and warranties contained in this Article V, neither Seller nor any other Person makes any other express or implied representation or warranty with respect to Seller, the Business, the Purchased Assets, the Assumed Liabilities or the transactions contemplated by this Agreement, and Seller disclaims any other representations or warranties, whether made by Seller, any Affiliate of Seller or any of their respective officers, directors, employees, agents or representatives. Except for the representations and warranties contained in Article V hereof (as modified by the Schedules hereto), Seller (i) expressly disclaims and negates any representation or warranty, expressed or implied, at common law, by statute, or otherwise, relating to the condition of the Purchased Assets (including any implied or expressed warranty of merchantability or fitness for a particular purpose, or of conformity to models or samples of materials) and (ii) disclaims all liability and responsibility for any representation, warranty, projection, forecast, statement, or information made, communicated, or furnished (orally or in writing) to Purchaser or its Affiliates or representatives (including any opinion, information, projection, or advice that may have been or may be provided to Purchaser by any director, officer, employee, agent, consultant, or representative of Seller or any of its Affiliates). Seller makes no representations or warranties to Purchaser regarding the probable success or profitability of the Business. The disclosure of any matter or item in any schedule hereto shall not be deemed to constitute an acknowledgment that any such matter is required to be disclosed or is material or that such matter would result in a Material Adverse Effect.

ARTICLE VI

REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to Seller that:

6.1 <u>Organization and Good Standing</u>. Each of the entities comprising Purchaser is a limited liability company duly organized, validly existing and in good standing in the State of its organization and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as now conducted.

6.2 <u>Authorization of Agreement</u>. Each of the entities comprising Purchaser has full corporate power and authority to execute and deliver this Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement or to be executed by Purchaser in connection with the consummation of the transactions contemplated hereby and thereby (the "<u>Purchaser Documents</u>"), and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by Purchaser of this Agreement and each Purchaser Document have been duly authorized by all necessary corporate action on behalf of the entities comprising Purchaser. This Agreement has been, and each Purchaser Document will be at or prior to the Closing, duly executed and delivered by Purchaser and (assuming the due

authorization, execution and delivery by the other parties hereto and thereto) this Agreement constitutes, and each Purchaser Document when so executed and delivered will constitute, the legal, valid and binding obligations of Purchaser, enforceable against Purchaser in accordance with their respective terms, subject to applicable bankruptcy, insolvency, reorganization, moratorium and similar laws affecting creditors' rights and remedies generally, and subject, as to enforceability, to general principles of equity, including principles of commercial reasonableness, good faith and fair dealing (regardless of whether enforcement is sought in a proceeding at law or in equity).

6.3 <u>Conflicts; Consents of Third Parties</u>.

(a) To the Seller's knowledge, none of the execution and delivery by Purchaser of this Agreement or the Purchaser Documents, the consummation of the transactions contemplated hereby or thereby, or the compliance by Purchaser with any of the provisions hereof or thereof will conflict with, or result in any violation of or default (with or without notice or lapse of time, or both) under, or give rise to a right of termination or cancellation under any provision of (i) the certificate of incorporation and by-laws of Purchaser, (ii) any Contract or Permit to which Purchaser is a party or by which Purchaser or its properties or assets are bound or (iii) any Order of any Governmental Body applicable to Purchaser or by which any of the properties or assets of Purchaser are bound or (iv) any applicable Law.

(b) Except for the Sale Order, no consent, waiver, approval, Order, Permit or authorization of, or declaration or filing with, or notification to, any Person or Governmental Body is required on the part of Purchaser in connection with the execution and delivery of this Agreement or the Purchaser Documents, the compliance by Purchaser with any of the provisions hereof or thereof, the consummation of the transactions contemplated hereby or thereby or the taking by Purchaser of any other action contemplated hereby or thereby, or for Purchaser to conduct the Business.

6.4 <u>Litigation</u>. There are no Legal Proceedings pending or, to the knowledge of Purchaser, threatened against Purchaser, or to which Purchaser is otherwise a party before any Governmental Body, which, if adversely determined, would reasonably be expected to have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or to consummate the transactions hereby. Purchaser is not subject to any Order of any Governmental Body except to the extent the same would not reasonably be expected to have a material adverse effect on the ability of Purchaser to perform its obligations under this Agreement or to consummate the transactions contemplated hereby.

6.5 <u>Financial Advisors</u>. No Person has acted, directly or indirectly, as a broker, finder or financial advisor for Purchaser in connection with the transactions contemplated by this Agreement and no Person is entitled to any fee or commission or like payment from Seller in respect thereof.

6.6 <u>Financial Capability</u>. Purchaser (i) has, and at the Closing will have, sufficient internal funds (without giving effect to any unfunded financing regardless of whether any such financing is committed) available to pay the Purchase Price and any expenses incurred by

Purchaser in connection with the transactions contemplated by this Agreement, (ii) has, and at the Closing will have, the resources and capabilities (financial or otherwise) to perform its obligations hereunder, and (iii) has not incurred any obligation, commitment, restriction or Liability of any kind, which would impair or adversely affect such resources and capabilities.

Condition of the Business. Notwithstanding anything contained in this 6.7 Agreement to the contrary, Purchaser acknowledges and agrees that Seller is not making any representations or warranties whatsoever, express or implied, beyond those expressly given by Seller in Article V hereof (as modified by the Schedules hereto as supplemented or amended), and Purchaser acknowledges and agrees that, except for the representations and warranties contained therein, the Purchased Assets and the Business are being transferred on a "where is" and, as to condition, "as is" basis. Any claims Purchaser may have for breach of representation or warranty shall be based solely on the representations and warranties of Seller set forth in Article V hereof (as modified by the Schedules hereto as supplemented or amended). Purchaser further represents that neither Seller nor any of its Affiliates nor any other Person has made any representation or warranty, express or implied, as to the accuracy or completeness of any information regarding Seller, the Business or the transactions contemplated by this Agreement not expressly set forth in this Agreement, and none of Seller, any of its Affiliates or any other Person will have or be subject to any liability to Purchaser or any other Person resulting from the distribution to Purchaser or its representatives or Purchaser's use of, any such information, including any confidential memoranda distributed on behalf of Seller relating to the Business or other publications or data room information provided to Purchaser or its representatives, or any other document or information in any form provided to Purchaser or its representatives in connection with the sale of the Business and the transactions contemplated hereby. Purchaser acknowledges that it has conducted to its satisfaction, its own independent investigation of the Business and, in making the determination to proceed with the transactions contemplated by this Agreement, Purchaser has relied on the results of its own independent investigation.

ARTICLE VII

BANKRUPTCY COURT MATTERS

7.1 <u>Competing Transaction</u>.

(a) This Agreement is subject to approval by the Bankruptcy Court and the consideration by Seller of higher or better competing offers (each a "<u>Competing Offer</u>"). From the date hereof (and any prior time) and until the transaction contemplated by this Agreement is consummated, Seller is permitted to cause its representatives and Affiliates to initiate contact with, solicit or encourage submission of any inquiries, proposals or offers by, any Person (in addition to Purchaser and its Affiliates, agents and representatives) in connection with any sale or other disposition of the Purchased Assets. In addition, Seller shall have the responsibility and obligation to respond to any inquiries or offers to purchase all or any part of the Purchased Assets and perform any and all other acts related thereto which are required under the Bankruptcy Code or other applicable law, including, without limitation, supplying information relating to the Business and the assets of Seller to prospective purchasers.

(b) In consideration of Purchaser conducting its due diligence and entering into this Agreement, and subject to the approval of the Bankruptcy Court as specified in <u>Section</u> <u>7.1(a)</u>, Seller agrees to pay Purchaser from the proceeds of any Competing Offer that are paid to and received by the Seller (i) a break up fee in an amount equal to 2% of \$51,250,000 (the "<u>Breakup Fee</u>"), and (ii) as reimbursement, all actual, necessary, reasonable and documented out-of-pocket expenses of Purchaser (the "<u>Expense Reimbursement</u>"); provided, however, that such Expense Reimbursement shall not exceed \$500,000.

7.2 <u>Bankruptcy Court Filings</u>. As more fully set forth in Section 8.12, Purchaser agrees that it will promptly take such actions as are reasonably requested by Seller to assist in obtaining entry of the Sale Order and a finding of adequate assurance of future performance by Purchaser, including furnishing affidavits or other documents or information for filing with the Bankruptcy Court for the purposes, among others, of providing necessary assurances of performance by Purchaser under this Agreement and demonstrating that Purchaser is a "good faith" purchaser under Section 363(m) of the Bankruptcy Code. Purchaser shall not, without the prior written consent of Seller, file, join in, or otherwise support in any manner whatsoever any motion or other pleading relating to the sale of the Purchased Assets hereunder. In the event the entry of the Sale Order or the Procedures Order shall be appealed, Seller and Purchaser shall use their respective reasonable efforts to defend such appeal, and Seller and Purchaser shall each be responsible solely for their own legal fees in such event.

ARTICLE VIII

COVENANTS

8.1 Access to Information. Seller agrees that, prior to the Closing Date, Purchaser shall be entitled, through its officers, employees and representatives (including, without limitation, its legal advisors and accountants), to make such investigation of the properties, businesses and operations of the Business and such examination of the books and records of the Business, the Purchased Assets and the Assumed Liabilities as it reasonably requests and to make extracts and copies of such books and records. Any such investigation and examination shall be conducted during regular business hours upon reasonable advance notice and under reasonable circumstances and shall be subject to restrictions under applicable Law. Seller shall cause the officers, employees, consultants, agents, accountants, attorneys and other representatives of Seller to cooperate with Purchaser and Purchaser's representatives in connection with such investigation and examination, and Purchaser and its representatives shall cooperate with Seller and its representatives and shall use their reasonable efforts to minimize any disruption to the Business. Notwithstanding anything herein to the contrary, no such investigation or examination shall be permitted to the extent that it would require to disclose information subject to attorney-client privilege or conflict with any confidentiality obligations to which Seller is bound.

8.2 <u>Conduct of the Business Pending the Closing</u>.

(a) Prior to the Closing, except (1) as set forth on <u>Schedule 8.2(a)</u>, (2) as required by applicable Law, (3) as otherwise expressly contemplated by this Agreement, or

(4) with the prior written consent of Purchaser (which consent shall not be unreasonably withheld or delayed), Seller shall:

(i) conduct the Business only in the Ordinary Course of Business;

(ii) use its commercially reasonable efforts to (A) preserve the present business operations, organization and goodwill of the Business, and (B) preserve the present relationships with customers and suppliers of the Business, and

(iii) refrain from entering into, modifying, or otherwise altering, any new or existing IP Contract without the prior written consent of Purchaser.

(b) From the date of execution of this Agreement until the Sale Termination Date (as defined in the Agency Agreement), Seller shall not terminate, reject, amend or modify any lease for a Store (as defined in the Agency Agreement) or contract related to the Business without the prior written consent of Purchaser(which consent shall not be unreasonably withheld or delayed).

8.3 <u>Consents</u>. Seller shall use its commercially reasonable efforts, and Purchaser shall cooperate with Seller, to obtain at the earliest practicable date all consents and approvals required to consummate the transactions contemplated by this Agreement, including, without limitation, the consents and approvals referred to in <u>Section 5.2(b)</u> hereof; <u>provided</u>, <u>however</u>, that Seller shall not be obligated to pay any consideration therefor to any third party from whom consent or approval is requested or to initiate any litigation or legal proceedings to obtain any such consent or approval.

8.4 <u>Regulatory Approvals</u>. Intentionally Omitted.

8.5 <u>Further Assurances</u>. Each of Seller and Purchaser shall use its commercially reasonable efforts to (i) take all actions necessary or appropriate to consummate the transactions contemplated by this Agreement and (ii) cause the fulfillment at the earliest practicable date of all of the conditions to their respective obligations to consummate the transactions contemplated by this Agreement.

8.6 <u>Confidentiality</u>. Purchaser acknowledges that the Confidential Information provided to it in connection with this Agreement, including under <u>Section 8.1</u>, and the consummation of the transactions contemplated hereby, is subject to those certain confidentiality agreements, dated April 28, 2008 and April 29, 2008 between the Purchaser and Seller (the "<u>Confidentiality Agreement</u>"), the terms of which are incorporated herein by reference. Effective upon, and only upon, the Closing Date, the Confidentiality Agreement shall terminate with respect to information relating solely to the Business or otherwise included in the Purchased Assets; <u>provided</u>, <u>however</u>, that Purchaser acknowledges that any and all other Confidential Information provided to it by Seller or its representatives concerning Seller shall remain subject to the terms and conditions of the Confidential Information" shall mean any confidential information with respect to, including, methods of operation, customers, customer lists, Products, prices, fees, costs, Technology, inventions, Trade Secrets, know-how, Software, marketing methods, plans, personnel, suppliers, competitors, markets or other specialized information or proprietary matters.

8.7 <u>Customer Lists</u>. Seller will take reasonable and necessary steps to ensure that the customer lists and contact information can be properly transferred.

8.8 <u>Preservation of Records</u>. Seller and Purchaser agree that each of them shall preserve and keep the records held by it or their Affiliates relating to the Business for a period of two (2) years from the Closing Date and shall make such records and personnel available to the other as may be reasonably required by such party in connection with, among other things, any insurance claims by, Legal Proceedings or tax audits against or governmental investigations of Seller or Purchaser or any of their Affiliates or in order to enable Seller or Purchaser to comply with their respective obligations under this Agreement and each other agreement, document or instrument contemplated hereby or thereby. In the event Seller or Purchaser wishes to destroy such records before or after that time, such party shall first give ninety (90) days prior written notice to the other and such other party shall have the right at its option and expense, upon prior written notice given to such party within such ninety (90) day period, to take possession of the records within one hundred and eighty (180) days after the date of such notice.

8.9 <u>Publicity</u>. Neither Seller nor Purchaser shall issue any press release or public announcement concerning this Agreement or the transactions contemplated hereby without obtaining the prior written approval of the other party hereto, which approval will not be unreasonably withheld or delayed, unless, in the sole judgment of Purchaser or Seller, disclosure is otherwise required by applicable Law or by the Bankruptcy Court with respect to filings to be made with the Bankruptcy Court in connection with this Agreement or by the applicable rules of any stock exchange on which Purchaser or Seller lists securities, <u>provided</u> that the party intending to make such release shall use its best efforts consistent with such applicable Law or Bankruptcy Court requirement to consult with the other party with respect to the text thereof.

8.10 <u>Use of Name</u>. Seller agrees that it shall (i) as soon as practicable after the Closing Date and in any event within sixty (60) days following the Closing Date, cease to make any use of the name "Sharper Image" or any service marks, trademarks, trade names, identifying symbols, logos, emblems, signs or insignia related thereto or containing or comprising the foregoing, including any name or mark confusingly similar thereto (collectively, the "<u>Seller</u> <u>Marks</u>"), (ii) immediately after the Closing, cease to hold itself out as having any affiliation with Purchaser or any of its Affiliates, and (iii) effect a change in the caption of its bankruptcy proceeding so that the words "Sharper Image" do not appear in such caption.

8.11 <u>Supplementation and Amendment of Schedules</u>. Seller may, at its option, include in the Schedules items that are not material in order to avoid any misunderstanding, and such inclusion, or any references to dollar amounts, shall not be deemed to be an acknowledgement or representation that such items are material, to establish any standard of materiality or to define further the meaning of such terms for purposes of this Agreement. Information disclosed in the Schedules shall constitute a disclosure for all purposes under this Agreement notwithstanding any reference to a specific section, and all such information shall be deemed to qualify the entire Agreement and not just such section. From time to time prior to the Closing, Seller shall have the right to supplement or amend the Schedules with respect to any matter hereafter arising or discovered after the delivery of the Schedules pursuant to this Agreement. No such supplement or amendment shall have any effect on the satisfaction of the condition to closing set forth in <u>Section 10.1(a)</u>; provided, however, if the Closing shall occur, then Purchaser shall be deemed to have waived any right or claim pursuant to the terms of this Agreement or otherwise, including pursuant to <u>Article XI</u> hereof, with respect to any and all matters disclosed pursuant to any such supplement or amendment at or prior to the Closing.

8.12 <u>Court Order</u>. Seller shall use commercially reasonable efforts to obtain the Sale Order. If a written objection is filed to the Sale Motion, which is an objection which would prohibit or otherwise prevent the Closing from occurring pursuant to the terms of this Agreement, Seller shall use commercially reasonable efforts to have such objection overruled.

Adequate Assurance of Future Performance. With respect to each Purchased 8.13 Contract, Purchaser shall use commercially reasonable efforts to provide adequate assurance as required under the Bankruptcy Code of the future performance of the applicable Purchased Contract by Purchaser. Purchaser agrees that it will promptly take all actions reasonably required by Seller or ordered by the Bankruptcy Court to assist in obtaining the Bankruptcy Court's entry of an order approving this Agreement, such as furnishing affidavits, nonconfidential financial information, confidential information subject to a reasonable form of confidentiality agreement or other documents or information for filing with the Bankruptcy Court and making Purchaser's employees and representatives available to be interviewed by Seller's attorneys and to testify before the Bankruptcy Court and at depositions, with respect to demonstrating adequate assurance of future performance by Purchaser under the Purchased Contract. If a written objection is filed to the Sale Motion, which is an objection which would prohibit or otherwise prevent the Closing from occurring pursuant to the terms of this Agreement, Purchaser shall use commercially reasonable efforts to have such objection overruled..

8.14 <u>Purchaser Covenants after Closing</u>. Purchaser covenants and agrees that it shall, from and after the Closing Date (unless otherwise agreed with Seller), do each of the following:

(a) <u>Access</u>. Purchaser shall, upon reasonable advance notice, afford to Seller's officers, independent public accountants, attorneys, consultants and other representatives, reasonable access during normal business hours to the Purchased Assets and all records pertaining to the Purchased Assets. To the extent that such Purchased Assets include information management systems, the Purchaser's officers, consultants and other representatives shall provide the Seller with access to such information management systems on a royalty-free basis solely for the purpose of enabling the Seller to conduct an orderly wind-down of the Seller's operations until such time as the wind-down is completed or December 31, 2008, whichever is earlier. Seller expressly acknowledges that nothing in this Section is intended to give rise to any contingency to Seller's obligations to proceed with the transactions contemplated herein; and

(b) <u>Contract Obligations</u>. Timely perform all obligations related to the Purchased Contracts and indemnify and hold harmless Seller and its officers, directors,

employees, agents, and representatives from and against all claims, demands, penalties, losses, liability or damage, including, without limitation, reasonable attorneys' fees and expenses, for or related to any failure to perform such obligations.

ARTICLE IX

EMPLOYEES AND EMPLOYEE BENEFITS

9.1 <u>Employment</u>. All matters relating to employees shall be governed by the Agency Agreement.

ARTICLE X

CONDITIONS TO CLOSING

10.1 <u>Conditions Precedent to Obligations of Purchaser</u>. The obligation of Purchaser to consummate the transactions contemplated by this Agreement and the Agency Agreement is subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions (any or all of which may be waived by Purchaser in whole or in part to the extent permitted by applicable Law):

(a) the representations and warranties of Seller set forth in this Agreement shall be true and correct at and as of the Closing, except to the extent such representations and warranties expressly relate to an earlier date (in which case such representations and warranties shall be true and correct on and as of such earlier date); <u>provided</u>, <u>however</u>, that in the event of a breach of a representation or warranty, the condition set forth in this <u>Section 10.1(a)</u> shall be deemed satisfied unless the effect of all such breaches of representations and warranties taken together result in a Material Adverse Effect, and Purchaser shall have received a certificate signed by an authorized officer of Seller, dated the Closing Date, to the foregoing effect;

(b) Seller shall have performed and complied in all material respects with all obligations and agreements required in this Agreement and the Agency Agreement to be performed or complied with by it prior to the Closing Date, provided, that the condition set forth in this <u>Section 10.1(b)</u> shall be deemed satisfied unless such failures to so perform or comply taken together result in a Material Adverse Effect, and Purchaser shall have received a certificate signed by an authorized officer of Seller, dated the Closing Date, to the forgoing effect; and

(c) Seller shall have delivered, or caused to be delivered, to Purchaser all of the items required by this Agreement.

10.2 <u>Conditions Precedent to Obligations of Seller</u>. The obligations of Seller to consummate the transactions contemplated by this Agreement and the Agency Agreement are subject to the fulfillment, prior to or on the Closing Date, of each of the following conditions (any or all of which may be waived by Seller in whole or in part to the extent permitted by applicable Law):

(a) the representations and warranties of Purchaser set forth in this Agreement and the Agency Agreement qualified as to materiality shall be true and correct, and those not so qualified shall be true and correct in all material respects, at and as of the Closing Date as though made on the Closing Date, except to the extent such representations and warranties relate to an earlier date (in which case such representations and warranties qualified as to materiality shall be true and correct, and those not so qualified shall be true and correct in all material respects, on and as of such earlier date), and Seller shall have received a certificate signed by an authorized officer of Purchaser, dated the Closing Date, to the foregoing effect;

(b) Purchaser shall have performed and complied in all respects with all obligations and agreements required by this Agreement and the Agency Agreement to be performed or complied with by Purchaser on or prior to the Closing Date, and Seller shall have received a certificate signed by an authorized officer of Purchaser, dated the Closing Date, to the foregoing effect; and

(c) Purchaser shall have delivered, or caused to be delivered, to Seller all of the items set forth in <u>Section 4.3</u>.

10.3 <u>Conditions Precedent to Obligations of Purchaser and Seller</u>. The respective obligations of Purchaser and Seller to consummate the transactions contemplated by this Agreement and the Agency Agreement are subject to the fulfillment, on or prior to the Closing Date, of each of the following conditions (any or all of which may be waived by Purchaser and Seller in whole or in part to the extent permitted by applicable Law):

(a) there shall not be in effect any Order by a Governmental Body of competent jurisdiction restraining, enjoining or otherwise prohibiting the consummation of the transactions contemplated hereby;

(b) the Bankruptcy Court shall have entered the Procedures Order, in form and substance reasonably acceptable to Seller and Purchaser, <u>provided</u> that the Procedures Order shall expressly approve the Breakup Fee and Expense Reimbursement as provided herein; and

(c) the Bankruptcy Court shall have entered the Sale Order and any stay period applicable to the Sale Order shall have expired or shall have been waived by the Bankruptcy Court, and such Sale Order shall be in full force and effect, and shall not have been modified, as of the Closing Date.

10.4 <u>Frustration of Closing Conditions</u>. Neither Seller nor Purchaser may rely on the failure of any condition set forth in <u>Section 10.1</u>, <u>10.2</u> or <u>10.3</u>, as the case may be, if such failure was caused by such party's failure to comply with any provision of this Agreement.

ARTICLE XI

NO SURVIVAL

11.1 <u>No Survival of Representations and Warranties</u>. The parties hereto agree that the representations and warranties contained in this Agreement shall not survive the Closing

hereunder, and none of the parties shall have any liability to each other after the Closing for any breach thereof. The parties hereto agree that the covenants contained in this Agreement to be performed at or after the Closing shall survive the Closing hereunder, and each party hereto shall be liable to the other after the Closing for any breach thereof.

11.2 <u>No Consequential Damages</u>. Notwithstanding anything to the contrary elsewhere in this Agreement and/or the Agency Agreement, no party shall, in any event, be liable to any other Person for any consequential, incidental, indirect, special or punitive damages of such other Person, including loss of future revenue, income or profits, diminution of value or loss of business reputation or opportunity relating to the breach or alleged breach hereof (provided that such limitation with respect to lost profits shall not limit Seller's right to recover contract damages in connection with Purchaser's failure to close in violation of this Agreement).

ARTICLE XII

TAXES

12.1 <u>Transfer Taxes</u>. The party who is liable under applicable law shall be responsible for (and shall indemnify and hold the other party harmless and its directors, officers, employees, Affiliates, agents, successors and permitted assigns against) any sales, use, stamp, documentary stamp, filing, recording, transfer or similar fees or taxes or governmental charges (including any interest and penalty thereon) payable in connection with the transactions contemplated by this Agreement ("<u>Transfer Taxes</u>"). Seller and Purchaser shall cooperate and consult with each other prior to filing any Tax Returns in respect of Transfer Taxes. Seller and Purchaser shall cooperate and otherwise take commercially reasonable efforts to obtain any available refunds for Transfer Taxes.

12.2 <u>Prorations</u>. All personal property Taxes or similar ad valorem obligations levied with respect to the Purchased Assets for any taxable period that includes the Closing Date and ends after the Closing Date, whether imposed or assessed before or after the Closing Date, shall be prorated between Seller and Purchaser as of 12:01 a.m. (Eastern time) on the Closing Date. If any Taxes subject to proration are paid by Purchaser, on the one hand, and Seller, on the other hand, the proportionate amount of such Taxes paid (or in the event of a refund of any portion of such Taxes previously paid is received, such refund) shall be paid promptly by (or to) the other after the payment of such Taxes (or promptly following the receipt of any such refund).

ARTICLE XIII

MISCELLANEOUS

13.1 <u>Expenses</u>. Except as otherwise provided in this Agreement, including, without limitation, in Section 7.1(b) hereof, and the Agency Agreement, each of Seller and Purchaser shall bear its own expenses incurred in connection with the negotiation and execution of this Agreement and each other agreement, document and instrument contemplated by this Agreement and the consummation of the transactions contemplated hereby and thereby.

13.2 <u>Injunctive Relief</u>. Damages at law may be an inadequate remedy for the breach of any of the covenants, promises and agreements contained in this Agreement and, accordingly, any party hereto shall be entitled to injunctive relief with respect to any such breach, including without limitation specific performance of such covenants, promises or agreements or an order enjoining a party from any threatened, or from the continuation of any actual, breach of the covenants, promises or agreements contained in this Agreement and the Agency Agreement. The rights set forth in this <u>Section 13.2</u> shall be in addition to any other rights which a Party may have at law or in equity pursuant to this Agreement and the Agency Agreement.

13.3 <u>Submission to Jurisdiction; Consent to Service of Process.</u>

(a) Without limiting any party's right to appeal any order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to enforce the terms of this Agreement and to decide any claims or disputes which may arise or result from, or be connected with, this Agreement and the Agency Agreement, any breach or default hereunder, or the transactions contemplated hereby, and (ii) any and all proceedings related to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the parties hereby consent to and submit to the jurisdiction and venue of the Bankruptcy Court and shall receive notices at such locations as indicated in Section 13.7 hereof; provided, however, that if the Bankruptcy Case has closed, the parties agree to unconditionally and irrevocably submit to the exclusive jurisdiction of the United States District Court for the Southern District of New York sitting in New York County or the Commercial Division, Civil Branch of the Supreme Court of the State of New York sitting in New York County and any appellate court from any thereof, for the resolution of any such claim or dispute. The parties hereby irrevocably waive, to the fullest extent permitted by applicable law, any objection which they may now or hereafter have to the laying of venue of any such dispute brought in such court or any defense of inconvenient forum for the maintenance of such dispute. Each of the parties hereto agrees that a judgment in any such dispute may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by law.

(b) Each of the parties hereto hereby consents to process being served by any party to this Agreement in any suit, action or proceeding by delivery of a copy thereof in accordance with the provisions of <u>Section 13.7</u>.

13.4 <u>Waiver of Right to Trial by Jury</u>. Each party to this Agreement waives any right to trial by jury in any action, matter or proceeding regarding this Agreement and the Agency Agreement or any provision hereof or therein.

13.5 <u>Entire Agreement; Amendments and Waivers</u>. This Agreement (including the schedules and exhibits hereto), and the Agency Agreement and the Confidentiality Agreement represent the entire understanding and agreement between the parties hereto with respect to the subject matter hereof. This Agreement and the Agency Agreement can be amended, supplemented or changed, and any provision hereof can be waived, only by written instrument making specific reference to this Agreement and the Agency Agreement signed by the party against whom enforcement of any such amendment, supplement, modification or waiver is sought. No action taken pursuant to this Agreement and the Agency Agreement, including without limitation, any investigation by or on behalf of any party, shall be deemed to constitute a

waiver by the party taking such action of compliance with any representation, warranty, covenant or agreement contained herein. The waiver by any party hereto of a breach of any provision of this Agreement shall not operate or be construed as a further or continuing waiver of such breach or as a waiver of any other or subsequent breach. No failure on the part of any party to exercise, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of such right, power or remedy by such party preclude any other or further exercise thereof or the exercise of any other right, power or remedy. All remedies hereunder are cumulative and are not exclusive of any other remedies provided by law.

13.6 <u>Governing Law</u>. This Agreement and the Agency Agreement shall be governed by and construed in accordance with the laws of the State of New York applicable to contracts made and performed in such State without regard to conflicts of laws principles thereof.

13.7 <u>Notices</u>. All notices and other communications under this Agreement shall be in writing and shall be deemed given (i) when delivered personally by hand (with written confirmation of receipt), (ii) when sent by facsimile (with written confirmation of transmission) or (iii) one business day following the day sent by overnight courier (with written confirmation of receipt), in each case at the following addresses and facsimile numbers (or to such other address or facsimile number as a party may have specified by notice given to the other party pursuant to this provision):

If to Seller, to:

Sharper Image Corporation 350 The Embarcadero, 6th Floor San Francisco, California, 94105 Attn: Robert Del Genio and Scott Silverman With a copy (which shall not constitute notice) to:

Weil, Gotshal & Manges LLP 767 Fifth Avenue New York, NY 10153 Facsimile: (212) 310-8007 Attn: Simeon Gold, Harvey R. Miller and Christopher Marcus

Cooley Godward Kronish LLP 1114 Avenue of the Americas New York, NY 10021 Facsimile: (212) 479-6275 Attn: Lawrence C. Gottlieb and Jay. R. Indyke If to Purchaser, to:

Hilco Trading Company, LLC One Northbrook Place 5 Revere Drive Northbrook, IL 60062 Attn: Joseph Malfitano, Esq. Tel: 847-504-3257 Fax: 847-897-0868

Gordon Brothers Group 101 Huntington Avenue - 10th Floor Boston, MA 02199 Attn: Rafael Klotz, Esq. Tel: 617-422-6246 Fax: 617-422-6288

With a copy to:

Morgan, Lewis & Bockius LLP 101 Park Avenue New York, NY 10178 Attn: Neil E. Herman, Esq. Tel: 212-309-6669 Fax: 212-309-6001

13.8 <u>Severability</u>. If any term or other provision of this Agreement and the Agency Agreement is invalid, illegal, or incapable of being enforced by any law or public policy, all other terms or provisions of this Agreement and the Agency Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any party. Upon such determination that any term or other provision is invalid, illegal, or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement and/or the Agency Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the greatest extent possible.

13.9 <u>Binding Effect; Assignment</u>. This Agreement and the Agency Agreement shall be binding upon and inure to the benefit of the parties and their respective successors and permitted assigns. Nothing in this Agreement shall create or be deemed to create any third party beneficiary rights in any Person or entity not a party to this Agreement except as provided below. No assignment of this Agreement or of any rights or obligations hereunder may be made by either Seller or Purchaser (by operation of law or otherwise) without the prior written consent of the other parties hereto and any attempted assignment without the required consents shall be void. No assignment of any obligations hereunder shall relieve the parties hereto of any such obligations. Upon any such permitted assignment, the references in this Agreement and/or the Agency Agreement to Purchaser shall also apply to any such assignee unless the context otherwise requires. In the event that a Chapter 11 trustee should be appointed for Seller, or in the event that Seller's Chapter 11 case should be converted to a case under Chapter 7, the obligations of Seller hereunder shall be binding upon such trustee or successor Chapter 7 estate.

13.10 <u>Conflict of Terms</u>. If any provision contained in this Agreement conflicts with any provisions contained in the Agency Agreement, the provision in the Agency Agreement shall govern.

13.11 <u>Non-Recourse.</u> No past, present or future director, officer, employee, incorporator, member, partner or equityholder of Seller shall have any liability for any obligations or liabilities of Seller under this Agreement, the Agency Agreement or the Seller Documents of or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby and thereby.

13.12 <u>Joint and Several Liability of the Purchaser</u>. Each of (a) Gordon Brothers Retail Partners, (b) GB Brands, LLC, (c) Hilco Merchant Resources, LLC, and (d) Hilco Consumer Capital, LLC shall be jointly and severally liable for the due and punctual payment of the Purchase Price and the performance of all other obligations of Purchaser set forth in this Agreement.

13.13 <u>Counterparts.</u> This Agreement and the Agency Agreement may be executed in as many counterparts as may be required, which counterparts may be delivered by facsimile or electronic mail, and it shall not be necessary that the signature of, or on behalf of, each party, appear on each counterpart; but it shall be sufficient that the signature of, or on behalf of, each party, or that the signatures of the persons required to bind any party, appear on one or more such counterparts. All such counterparts when taken together shall constitute a single and legally binding agreement.

[REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK SIGNATURE PAGES TO FOLLOW]

IN WITNESS WHEREOF, the parties hereto have caused this Agreement to be executed by their respective officers thereunto duly authorized, as of the date first written above.

SHARPER IMAGE CORPORATION

By: <u>/s/ Rebecca L. Roedell</u> Name: Rebecca L. Roedell Title: Executive Vice President & Chief Financial Officer

HILCO MERCHANT RESOURCES, LLC,

By: <u>/s/ Eric W. Kaup</u> Name: Eric W. Kaup Title: Executive Vice President, General Counsel, Member

HILCO CONSUMER CAPITAL, LLC,

By: <u>/s/ Eric W. Kaup</u> Name: Eric W. Kaup Title: Executive Vice President, General Counsel, Managing Member

GORDON BROTHERS RETAIL PARTNERS, LLC,

By: /s/ Rafael Klotz Name: Rafael Klotz Title: Managing Director

GB BRANDS, LLC,

By: /s/ Rafael Klotz Name: Rafael Klotz Title: Managing Director

Schedule 2.6(b)

- 1. License Agreement dated March 11, 2008 by and between Sharper Image Corporation and ARG Manufacturing Inc.
- 2. License Agreement dated October 23, 2007 by and between Sharper Image Corporation and Diversified Computer Supplies Inc.
- 3. License Agreement dated December 10, 2007 by and between Sharper Image Corporation and E&B Giftware LLC.
- 4. License Agreement dated June 27, 2007 by and between Sharper Image Corporation and ENE Group LLC.
- 5. License Agreement dated June 8, 2007 by and between Sharper Image Corporation and Futai USA Inc.
- 6. License Agreement dated August 9, 2007 by and between Sharper Image Corporation and LG International (America) Inc.
- 7. License Agreement dated February 29, 2008 by and between Sharper Image Corporation and MerchSource, LLC.
- 8. License Agreement dated March 1, 2006 by and between Sharper Image Corporation and OfficeMax Incorporated.
- 9. Amendment No. 1 to Trademark License Agreement dated August 17, 2007 by and between Sharper Image Corporation and OfficeMax Incorporated.
- 10. License Agreement dated October 23, 2007 by and between Sharper Image Corporation and Polaris Media Research Inc d/b/a Moneaul Lab.
- 11. License Agreement dated September 1, 2007 by and between Sharper Image Corporation and PureHiTech, Inc.
- 12. License Agreement dated October 10, 2007 by and between Sharper Image Corporation and Rilliant Systems Inc.
- 13. License Agreement dated May 1, 2007 by and between Sharper Image Corporation and Sensio Inc.
- 14. Amendment No. 1 to Sensio Inc. License Agreement dated March 1, 2008 by and between Sharper Image Corporation and Sensio Inc.

- 15. Amendment No. 2 to Sensio Inc. License Agreement dated March 1, 2008 by and between Sharper Image Corporation and Sensio Inc.
- 16. License Agreement dated April 14, 2008 by and between Sharper Image Corporation and Icon Eyewear, Inc.

AGENCY AGREEMENT

This Agency (the "<u>Agreement</u>") is made as of May 13, 2008, by and between a joint venture formed among: (a) Hilco Consumer Capital, LLC, (b) Hilco Merchant Resources, LLC, (c) Gordon Brothers Retail Partners, LLC, and (d) GB Brands, LLC (collectively, the "<u>Agent</u>") and Sharper Image Corporation (the "<u>Merchant</u>"), with a principal place of business at 350 The Embarcadero, 6th floor, San Francisco, CA 94105.

RECITALS

WHEREAS, the Merchant has commenced a case (the "<u>Bankruptcy Case</u>") under chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>") in the United States Bankruptcy Court for the District of Delaware (the "<u>Bankruptcy Court</u>");

WHEREAS, the Merchant operates retail stores and desires that the Agent act as the Merchant's exclusive agent for the limited purpose of selling all of the Merchandise located in the Merchant's retail store location(s) (each a "<u>Store</u>" and, collectively, the "<u>Stores</u>"), and certain merchandise located in Merchant's distribution centers all listed on <u>Exhibit 1</u> attached hereto, by means of a "going-out-of-business," "store closing," "bankruptcy," or similarly themed sale (as further described below, the "<u>Sale</u>");

WHEREAS, the Agent has agreed to act as the Merchant's exclusive agent on the terms and conditions set forth herein;

WHEREAS, contemporaneously herewith, the Agent and the Merchant have entered into that certain Asset Purchase Agreement, dated as of May ____, 2008 (the "<u>APA</u>"), which is hereby incorporated herein.

NOW THEREFORE, in consideration of the mutual covenants and agreements set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged, the Agent and the Merchant hereby agree as follows:

Section 1. <u>Defined Terms</u>. All capitalized terms not defined herein shall have the meaning set forth in the APA. As used herein, the following terms shall have the following meanings:

"Agreement" has the meaning assigned to that term in the Preamble hereto.

"Agent" has the meaning assigned to that term in the Preamble hereto.

"APA" has the meaning assigned to that term in the Recitals hereto.

"Bankruptcy Case" has the meaning assigned to that term in the APA.

"Bankruptcy Code" has the meaning assigned to that term in the APA.

"Benefits Cap" has the meaning assigned to that term in Section 4.1 hereof.

"Central Expenses" has the meaning assigned to that term in Section 4.2 hereof.

"Closing" means the closing of transactions contemplated by this Agreement and the APA as set forth in Section 2.3 hereof.

"COBRA" means the Consolidated Omnibus Budget Reconciliation Act of 1986, as amended.

"Committee" means the statutory creditors' committee appointed in the Bankruptcy Case.

"Defective Merchandise" has the meaning assigned to that term in Section 5.1(b)

hereof.

"Designated Merchant Account" has the meaning assigned to that term in Section 3.1 hereof.

"DIP Credit Agreement" has the meaning assigned to that term in Section 3.3 hereof.

"Environmental Law" has the meaning assigned to that term in Section 11.1(o) hereof.

	"Excluded Benefits" has the meaning assigned to that term in Section 4.2 hereof.
	"Excluded Goods" has the meaning assigned to that term in Section 5.3 hereof.
	"Expense L/C" has the meaning assigned to that term in Section 3.3 hereof.
	"Expenses" has the meaning assigned to that term in Section 4 hereof.
	"FF&E" has the meaning assigned to that term in Section 5.1 hereof.
	"Final Inventory Report" has the meaning assigned to that term in Section 3.3(b)
hereof.	
	"Final Reconciliation" has the meaning assigned to that term in Section 3.4
hereof.	
	"Guaranteed Amount" has the meaning assigned to that term in Section 3.1
hereof.	
	"Guaranty L/C" has the meaning assigned to that term in Section 3.3 hereof.
	"Gross Rings" has the meaning assigned to that term in Section 5.5 hereof.
	"Inventory Date' has the meaning assigned to that term in Section 5.1 hereof.
	"Initial Guaranty Payment" has the meaning assigned to that term in Section 3.3
hereof.	
	"Inventory Taking" has the meaning assigned to that term in Section 5.4 hereof.
	"Inventory Taking Service" has the meaning assigned to that term in Section 5.4
hereof.	
	"Lender" has the meaning assigned to that term in Section 3.3 hereof.

"Lien" means any lien, security interest, pledge, hypothecation, encumbrance or other interest or claim (including but not limited to any and all "claims," as defined in section 10l(5) of the Bankruptcy Code, and any and all rights and claims under any bulk transfer statutes and similar laws) in or with respect to any of the Leases (including but not limited to any options or rights to purchase or recapture such Leases and any mechanics' or tax liens), whether arising by agreement, by statute or otherwise and whether arising prior to, on or after the Petition Date.

"Maximum Inventory Amount" has the meaning assigned to that term in Section 3.1 hereof.

"Merchandise" has the meaning assigned to that term in Section 5.1 hereof.

"Merchant" has the meaning assigned to that term in the Preamble hereto.

"Minimum Inventory Amount" has the meaning assigned to that term in Section

3.1 hereof.

"Occupancy Expenses" has the meaning assigned to that term in Section 3.3 hereof.

"On-Order Merchandise has the meaning assigned to that term in Section 5.1(b) hereof.

"Performance Bonuses" has the meaning assigned to that term in Section 9.4 hereof.

"Petition Date" means February 19, 2008.

"Proceeds" has the meaning assigned to that term in Sections 3.1 and 7.1 hereof.

"Remaining Merchandise" has the meaning assigned to that term in Section 3.2

hereof.

"Retained Employee" has the meaning assigned to that term in Section 9.1 hereof.

"Retail Price" has the meaning assigned to that term in Section 5.2 hereof.

"Sale" has the meaning assigned to that term in the Recitals hereto.

"Sale Commencement Date" has the meaning assigned to that term in Section 6.1 hereof.

"Sale Documents" has the meaning assigned to that term in Section 12.1 hereof. "Sale Order" has the meaning assigned to that term in Section 2.2 hereof. "Sale Term" has the meaning assigned to that term in Section 6.1 hereof. "Sale Termination Date" has the meaning assigned to that term in Section 6.1

hereof.

"Sales Taxes" has the meaning assigned to that term in Section 8.3 hereof.

"Store" has the meaning assigned to that term in the Recitals hereto.

"Supplies" has the meaning assigned to that term in Section 8.4 hereof.

"Taxes" means (i) all federal, state, local and foreign taxes, including income, gross receipts, excise, employment, sales, use, transfer, license, payroll, franchise, severance, stamp, withholding, social security, unemployment, disability, real property, personal property, registration, environmental, custom duties, value added, alternative or add-on minimum, estimated or other tax, including any interest, penalties or additions thereto, whether disputed or not, or (ii) liability for the payment of any amounts of the type described in clause (i) as a result of being party to any agreement or any express or implied obligation to indemnify any other person, or (iii) in the case of Seller, liability for the payment of any amount of the type described in clause (i) as a result of being or having been before the Closing Date a member of an affiliated, consolidated, combined or unitary group, or a party to any agreement or arrangement, as a result of which liability of Seller is determined or taken into account with reference to the liability of any other person.

"Vacate Date" has the meaning assigned to that term in Section 6.2 hereof.

"WARN Act" means the Workers' Adjustment and Retraining Notification Act.

Section 2. <u>Appointment of Agent</u>.

2.1. The Merchant hereby appoints the Agent, and the Agent hereby agrees to serve, as the Merchant's exclusive agent for the purposes conducting the Sale in accordance with the terms and conditions of this Agreement.

2.2. The Merchant shall apply to the Bankruptcy Court for an order approving this Agreement and the APA in their entirety and the transactions contemplated therein (the "Sale Order"). The Sale Order shall provide, in a form reasonably satisfactory to the Merchant and Agent, inter alia, that (i) this Agreement and the APA (and each of the transactions contemplated thereby) are approved in their entirety; (ii) Merchant and Agent shall be authorized to continue to take any and all actions as may be necessary or desirable to implement this Agreement and the APA and each of the transactions contemplated thereby; (iii) Agent shall be entitled to sell all Merchandise hereunder free and clear of all liens, claims or encumbrances thereon, with any presently existing liens encumbering all or any portion of the Merchandise or the Proceeds attaching only to the Guaranteed Amount and other amounts to be received by Merchant under this Agreement; (iv) Agent shall have the right to use the Stores and all related Store services, furniture, fixtures, equipment and other assets of Merchant as designated hereunder for the purpose of conducting the Sale, free of any interference from any entity or person subject to compliance with the Sale Order; (v) Agent, as agent for Merchant, is authorized to conduct, advertise, post signs and otherwise promote the Sale as a "going-out-of-business," "store closing," "bankruptcy," or similarly themed sale (including, without limitation, by means of media advertising, A-frame, interior and exterior banners and similar signage and use of sign walkers), without further consent of any person, in accordance with the terms and conditions of this Agreement and the Sale Order (as the same may be modified and approved by the Bankruptcy Court), and without further compliance with applicable federal, state or local laws governing, *inter alia*, the conduct of store closing sales (the "Liquidation Sale Laws"), other than those designed to protect public health and safety; (vi) Agent shall be granted a limited license and right to use until the Sale Termination Date the trade names, logos and customer and mailing lists relating to and used in connection with the operation of the Stores, solely for the purpose of advertising the Sale in accordance with the terms of this Agreement; (vii) all utilities, landlords, creditors and all persons acting for or on their behalf shall not interfere with or otherwise impede the conduct of the Sale, institute any action in any court (other than in the Bankruptcy Court) or before any administrative body which in any way directly or indirectly interferes with or obstructs or impedes the conduct of the Sale; (viii) the Bankruptcy Court shall retain jurisdiction over the parties to enforce this Agreement and the APA; (ix) Agent shall not be liable for any claims against the Merchant other than as expressly provided for in this Agreement; and (x) Agent shall be granted a security interest in the Merchandise and the Proceeds as provided for in Section 16 hereof.

2.3. <u>Closing</u>. If the Sale Order is entered, then, subject to the satisfaction or waiver by the parties of the conditions to their respective obligations set forth in Sections 11 and 12 below and in the APA, the Closing shall take place as set forth in the APA. All transactions at the Closing shall be deemed to take place simultaneously and none shall be deemed to have taken place until all shall have taken place.

2.4. <u>Assumed Liabilities</u>. Except as specified in this Agreement and the APA, the Agent does not assume any liabilities or obligations of the Merchant, including, without

limitation, severance, termination pay, pension, profit sharing or any other employee benefit plans, compensation or retiree medical and other benefits and obligations or any obligation, claim or amount under the WARN Act or COBRA, Taxes, Liens, "adequate protection" obligations, mortgage obligations or legal fees incurred by professionals retained in the Merchant's Bankruptcy Case, liability to any landlords under the Leases for reimbursement of prior construction work, rent concession, allowances or the like, whether known or unknown, disputed or undisputed, contingent or non-contingent, liquidated or unliquidated, or otherwise.

Section 3. <u>Guaranteed Amount and Other Payments</u>.

3.1. <u>Payments to the Merchant and the Agent</u>.

(a) As a guaranty of the Agent's performance hereunder, and under the contemporaneously entered APA, in addition to the payment of Expenses as provided for in Section 4 below and subject to the potential adjustments in this Agreement and the APA, the Agent guarantees that the Merchant shall receive the Purchase Price (as defined in the APA) (for purposes of this Agreement, the "<u>Guaranteed Amount</u>"). The Agent shall pay to the Merchant the Guaranteed Amount, as adjusted in part, to reflect the values calculated through the: (A) Final Inventory Report, and (B) aggregate amount of Gross Rings, as adjusted for shrinkage per this Agreement.

(b) All proceeds of the Sale (the "<u>Proceeds</u>") shall be deposited in the Merchant's existing accounts and disbursed in accordance with Section 3.3 hereof; <u>provided that</u> as soon as practicable following the Sale Commencement Date, the Merchant shall establish segregated accounts for deposit of the Proceeds (the "<u>Designated Merchant Accounts</u>"). The Merchant shall pay to the Agent all Proceeds on a daily basis, commencing on the first business

day following payment of the Initial Guaranty Payment and delivery of the Guaranty L/C and Expense L/C.

(c) The Guaranteed Amount has been calculated and agreed to based upon the aggregate Retail Price of the Merchandise not being less than, or in excess of, \$56,000,000 (the "<u>Minimum Inventory Amount</u>"). To the extent that the aggregate Retail Price of the Merchandise, as determined by the final results of the Inventory Taking (defined below), is (x) less than the Minimum Inventory Amount, then the Guaranteed Amount shall be decreased by an amount equal to 65% of such deficiency, or (y) greater than the Minimum Inventory Amount, the Guaranteed Amount shall be increased by an amount equal to 36% of such excess.

(d) On the Sale Commencement Date, Merchant shall provide \$1,000 in cash in each of the Stores. Within 10 days of the Sale Commencement Date, Agent shall reimburse Merchant for all of such cash. Merchant and Agent shall cooperate to develop mutually agreeable procedures to verify such amounts.

3.2. <u>Remaining Merchandise</u>. All unsold Merchandise remaining, if any, in the Stores at the Sale Termination Date (the "<u>Remaining Merchandise</u>") shall become the property of the Agent, free and clear of all liens, claims and encumbrances.

3.3. <u>Time of Payments and Control of Proceeds.</u>

(a) On the Closing Date, the Agent shall (i) pay the Merchant ninety percent (95%) of the Guaranteed Amount (the "<u>Initial Guaranty Payment</u>"), (ii) pay the first week's estimated Expenses as mutually agreed to by the Merchant and the Agent, and (iii) deliver to Wells Fargo Retail Finance, LLC (the "<u>Lender</u>"), as agent and lender under that certain Debtor in Possession Loan and Security Agreement, dated February 19, 2008 (the "<u>DIP Credit</u> <u>Agreement</u>"), as beneficiary, two letters of credit from a financial institution acceptable to the

Merchant as follows: (A) a "<u>Guaranty L/C</u>" in the original face amount equal to the unpaid portion of the Guaranteed Amount, in a form to be agreed upon by the Merchant, the Agent, and the Lender, which shall terminate upon payment of the full Guaranteed Amount, and (B) an "<u>Expense L/C</u>" in the original face amount equal to three (3) weeks of estimated Expenses as mutually agreed to by the Merchant and the Agent, in a form to be agreed to by the Merchant, the Agent, and the Lender. The Initial Guaranty Payment and amount of the Guaranty L/C shall be calculated based upon the estimated aggregate Retail Price of the Merchandise to be included in the Sale, as reflected on the Merchant's books and records on the last business day immediately preceding the Sale Commencement Date minus the aggregate Retail Price for the On-Order Merchandise. Payments related to On-Order Merchandise shall be made as part of the weekly Sale reconciliation as On-Order Merchandise is received at the Stores.

(b) The unpaid portion of the Guaranteed Amount shall be paid by the Agent to the Merchant on the later of (i) the date that is 30 days after the Sale Commencement Date, and (ii) the second business day following the issuance of the final audited report of the aggregate Retail Price of the Merchandise by the Inventory Taking Service, after verification and reconciliation of the listing and tabulation of the Inventory Taking by the Agent and the Merchant (the "<u>Final Inventory Report</u>"). In the event there is any dispute with respect to the reconciliation of the aggregate Retail Price of the Merchandise following the Inventory Taking, then any such dispute shall be resolved by the Bankruptcy Court.

(c) All amounts required to be paid by the Agent or the Merchant under any provision of this Agreement shall be made by wire transfer of immediately available funds, which shall be wired by the Agent or the Merchant, as applicable, no later than 2:00 p.m. (Eastern Time) on the date that such payment is due. In the event that the date on which such payment is due is not a business day, then such payment shall be due on the next business day.

(d) The Guaranty L/C and the Expense L/C shall secure payment of the balance of any unpaid portion of the Guaranteed Amount, the Expenses, and any other amounts due from the Agent to the Merchant hereunder, shall be issued by a bank selected by the Agent and reasonably acceptable to the Merchant, and shall contain terms, provisions and conditions mutually acceptable to the Merchant and the Agent. Within five (5) business days after its receipt of written notice from the Lender or the Merchant that the obligations owed to the Lender have each been paid in full and all commitments under the DIP Credit Agreement terminated, the Agent shall cause the Guaranty L/C and the Expense L/C to be amended to name the Merchant as the beneficiary. The Guaranty L/C and the Expense L/C shall expire no earlier than 60 days after the Sale Termination Date. Unless the parties have mutually agreed in writing that they have completed the Final Reconciliation under this Agreement, then, at least 30 days prior to the expiration date of the Guaranty L/C and the Expense L/C, the Agent shall amend the Guaranty L/C and the Expense L/C solely to extend the respective expiration dates by at least an additional 60 days. If the Lender or the Merchant, as the case may be, does not receive such amendments to the Guaranty L/C and the Expense L/C, then all amounts hereunder shall become immediately due and payable, and the Lender or the Merchant, as the case may be, shall be permitted to draw the full amounts available under the Guaranty L/C and the Expense L/C and apply the amounts so drawn towards amounts then owed. Any amounts drawn in excess of the amounts then owed shall be held as security for amounts that may thereafter become due and payable to the Merchant hereunder.

In the event that the Agent, after receipt of five (5) days' written (e) notice (which notice shall not be required if the Agent or any member of the Agent shall be a debtor under the Bankruptcy Code), fails to pay the Guaranteed Amount, Expenses, or any portion thereof, or other obligations hereunder when due, the Lender or the Merchant, as the case may be, shall be permitted to draw the full amounts available under the Guaranty L/C and the Expense L/C and apply the amounts so drawn in payment of amounts then owed, and any amounts drawn in excess of the amounts then owed shall be held as security for amounts that may thereafter become due and payable to the Merchant hereunder. To the extent that after drawing upon the Guaranty L/C and the Expense L/C there are additional unpaid amounts due to the Merchant or the Lender, the Agent shall immediately pay such additional amounts in accordance with Section 3.3(c) hereof. The Merchant and the Agent agree that after payment of the full Guaranteed Amount (calculated pursuant to the Final Inventory Report) pursuant to Section 3.3(b) hereof, the Guaranty L/C shall be terminated and redelivered to the Agent. The remedies of the Merchant and the Lender under this Section 3.3(e) are in addition to any other remedies the Merchant and the Lender may have at law, equity, or under this Agreement, including without limitation, Section 3.4 hereof.

(f) During the Sale Term, all Proceeds of the Sale (including credit card proceeds), shall be deposited on a daily basis into the Designated Merchant Accounts. Commencing on the first business day following the payment of the Initial Guaranty Payment and the posting of the Guaranty L/C and the Expense L/C, and on each business day thereafter (or as soon thereafter as is practicable), the Merchant shall promptly pay to the Agent by wire funds transfer all collected funds constituting Proceeds of the Sale held by the Merchant (including credit card Proceeds, but excluding any other funds, including, without limitation, any

proceeds of Merchant's inventory sold prior to the Sale Commencement Date, such as special order goods or collections of accounts receivable at the Store level, if any).

(g) Merchant agrees that if at any time during the Sale Term Merchant holds any amounts due to Agent as Proceeds hereunder, Agent may, in its discretion, offset such Proceeds being held by Merchant against any amounts due and owing to Merchant pursuant to this Section 3.3 or otherwise under this Agreement. In addition, Merchant and Agent further agree that if at any time during the Sale Term Agent holds any amounts due to Merchant under this Agreement, Agent may, in its discretion, offset such amounts being held by it against any amounts due and owing by, or required to be paid by, Merchant hereunder.

(h) If and to the extent that Agent over-funds any amounts in respect of the Guaranteed Amount hereunder, then Merchant agrees to promptly reimburse such undisputed overpayment amounts to Agent. To the extent that any over-funded amounts in respect of the Guaranteed Amount hereunder have been received by the Lender and have not been reimbursed by Merchant, Agent shall inform the Lender of such overpayment in respect of the Guaranteed Amount hereunder and the Lender agrees to disgorge such overpayment in respect of the Guaranteed Amount hereunder to Agent within two (2) business days of such notice.

3.4. <u>Final Reconciliation</u>.

(a) Within thirty (30) days after the end of the Sale Term, Agent and Merchant shall complete a final reconciliation of the Sale, the written results of which shall be certified by representatives of each of Merchant and Agent as a final settlement of accounts between Merchant and Agent (the "<u>Final Reconciliation</u>"). Within five (5) days of completion of the Final Reconciliation, the Agent shall pay to the Merchant, or the Merchant shall pay to the Agent, as the case may be, any and all amounts due the other pursuant to the Final Reconciliation. During the Sale Term, and until all of the Agent's obligations under this Agreement have been satisfied, the Merchant and the Agent shall have reasonable access to the Merchant's and the Agent's records with respect to taxes and Expenses to review and audit such records.

(b) In the event that there is any dispute with respect to the Final Reconciliation, such dispute shall be promptly (and in no event later than the third business day following the written request by either the Merchant or the Agent) submitted to the Bankruptcy Court for a determination. The Merchant and the Agent hereby agree to submit to the jurisdiction of the Bankruptcy Court for such determination.

Section 4. <u>Expenses of the Sale</u>.

4.1. <u>Expenses</u>. The Merchant is obligated to pay, subject to its right to receive reimbursement on a weekly basis pursuant to the reconciliation procedures set forth in Section 4.3 below, the expenses directly incurred in connection with and attributable to the Sale (collectively, the "<u>Expenses</u>"), limited to the following:

(a) (i) base payroll of the Merchant's Store-level Retained Employees used in connection with the Sale for actual days worked (or in the case of hourly employees, the hours worked), plus (ii) an amount not to exceed **21.5% percent** of such base payroll (the "<u>Benefits Cap</u>") for the payment of all actual related payroll taxes, workers' compensation, and benefits of the Merchant's Retained Employees used in connection with the Sale (including, without limitation, medical and dental benefits, group life insurance, accidental death and dismemberment insurance, short and long term disability, and accrual for vacation and holiday

pay) for all such Retained Employees used, which accrue during the period of the Sale and are attributable to the Sale, plus (iii) actual costs payable to third party payroll processors;

(b) costs of all security in the Stores including, without limitation, courier and guard service;

(c) Performance Bonuses for Retained Employees, plus payroll taxes, as provided for in Section 9.4 below;

(d) (i) advertising and direct mailings relating to the Sale and (ii) Store interior and exterior signage and banners;

(e) local and long-distance telephone expenses and line charges;

(f) bank card fees, bank card error fees, credit card fees, and chargebacks in respect of disputed sales;

(g) bank service charges, check guarantee fees, and bad check expenses, each as relates to the Sale;

(h) costs for additional Supplies in accordance with Section 8.4 hereof;

(i) all fees and charges required to comply with all laws and regulations applicable to the Sale;

(j) any and all costs, including delivery and freight costs, related to the processing, transfer and consolidation of Merchandise between the Stores and the distribution centers (excluding all costs, including delivery and freight costs, to deliver the On-Order Merchandise to the Stores, but including, from and after the Sale Commencement Date, all costs of processing the On-Order Merchandise upon its arrival at the Stores and thereafter);

(k) trash removal, housekeeping and cleaning expenses related to the Stores;

(1) all reasonable travel expenses, including living expenses, payable to Merchant's Retained Employees relating to travel by such Retained Employees at the direction of the Agent, which shall include, without limitation, the costs of transferring the Merchant's Retained Employees between Stores;

(m) all costs and expenses of providing such additional Store-level services, including, without limitation, the employment of temporary help (which shall be coordinated and implemented through the Merchant's human resources department), which the Agent in its reasonable discretion considers appropriate, and other approved miscellaneous Store-level expenses incurred by the Agent;

(n) postage, courier and overnight mail charges to and from or amongthe Stores and central office (solely to the extent relating to the Sale);

(o) Occupancy Expenses on a per Store, per diem, basis and limited to those categories of expenses described and up to the maximum amounts set forth on **Exhibit 4.1** hereto;

(p) Central Expenses of \$50.00 per Store per week during the SaleTerm (prorated for any partial week);

(q) the Merchant's liability insurance and casualty insurance premiums in the amount of \$25.00 per Store per week (prorated for any partial week) required under Sections 13.1 and 13.2 hereto;

(r) any other expense approved by the Merchant directly incurred by the Agent in connection with the Sale;

(s) Agent's documented cost of capital (which documents shall be provided to a representative of the Merchant agreeable to the Merchant and the Agent and subject to appropriate confidentiality agreement), including letter of credit fees;

(t) Agent's reasonable supervision expense, including, (i) fees of the Agent's supervisors, (ii) expenses and travel costs for Agent's supervisors, and (iii) supervisor bonuses;

(u) Agent's due diligence and legal expenses ;

(v) 50% of the fees, costs and expenses of the Inventory Taking Service for the Inventory Taking; provided that the fees, costs, and expenses of the Inventory Taking Service for the Inventory Taking shall not include any fees, costs, or expenses relating to Retained Employees utilized during the Inventory Taking, which expenses shall be borne solely by Agent;

(w) (w) Agent's actual cost of insurance as required by Section 13.3

hereof.

There will be no double payment of Expenses to the extent that an Expense appears or is contained in more than one category. Notwithstanding anything herein to the contrary, to the extent that any Expense listed in Section 4.1 is also included on Exhibit 4.1, then Exhibit 4.1 shall control and such Expenses shall not be double counted.

4.2. <u>Definition of Certain Expenses</u>. As used herein, the following terms have the following respective meanings:

(a) "Central Expenses" means costs and expenses for the Merchant's central administrative services necessary for the Sale consisting of sales audit, MIS services,

POS systems, payroll processing, cash reconciliation, inventory processing and handling, data processing and reporting and any similar services.

(b) "Excluded Benefits" means the following benefits, except as provided in Section 4.1(a): maternity leave or other leaves of absence, termination or severance pay, union dues or other amounts due under any union contract or collective bargaining agreement, pension benefits, ERISA coverage and similar contributions, and payroll taxes, worker's compensation, and health insurance benefits, and any other benefits in excess of the Benefits Cap, including any payments due under the Worker Adjustment Retraining Notification Act ("<u>WARN Act</u>").

(c) "Occupancy Expenses" means base rent, percentage rent, HVAC, utilities, CAM, real estate and use taxes, merchant's association dues and expenses, rental for furniture, fixtures and equipment (including, without limitation, Store-level point of sale equipment), cash register maintenance, security systems, building alarm service, alarm service maintenance and store trash and snow removal expenses, all of the foregoing as categorized or reflected on **Exhibit 4.1** hereto.

(d) "Expenses" shall not include: (i) Excluded Benefits; (ii) Central Expenses in excess of the amount set forth in Section 4.1(p) above; and (iii) any other costs, expenses or liabilities payable by the Merchant under this Agreement (other than Expenses or other obligations of the Agent hereunder) all of which shall be paid by Merchant promptly when due during the Sale Term.

4.3. <u>Payment of Expenses</u>. The Agent shall be responsible to reimburse the Merchant for all Expenses of the Sale. All Expenses incurred during each week of the Sale (i.e., Sunday through Saturday) shall be paid by the Merchant as provided for herein, subject to

reimbursement by the Agent at the end of each week based on a weekly estimate to be mutually agreed upon on a weekly basis by the Merchant and the Agent or based upon an invoice submitted by the Merchant to the Agent. The Agent and the Merchant may review or audit the Expenses at any time.

Section 5. <u>Inventory Valuation; Merchandise</u>.

5.1. Merchandise Subject to this Agreement.

(a) For purposes of this Agreement, "Merchandise" shall mean: (i) all saleable finished goods inventory that is owned by the Merchant and located at the Stores as of the Sale Commencement Date (including saleable samples, open box items, in-store clearance items, discontinued items and returns); (ii) Defective Merchandise; and (iii) On-Order Merchandise. Notwithstanding the foregoing, "Merchandise" shall not include: (1) goods which belong to sublessees, licensees, department lessees, or concessionaires of Merchant; (2) goods held by Merchant on memo, on consignment, or as bailee; (3) furnishings, trade fixtures, equipment and/or improvements to real property which are located in the Stores (collectively, "<u>FF&E</u>"); (4) Excluded Defective Merchandise; (5) Merchant Consignment Goods; and (6) Additional Agent Merchandise.

(b) As used in this Agreement the following terms have the respective meanings set forth below:

(i) "Defective Merchandise" means any goods reasonably agreed upon by the Merchant's representative and the Agent's representative during the Inventory Taking as defective or otherwise not saleable in the ordinary course of Merchant's business at Retail Price, as if the Merchandise were not defective and as to which Merchant and Agent mutually agree as to the Retail Price of such Defective Merchandise. Display merchandise shall not be presumed to be defective. Packaging material that has minor wear, soiling, stains, dents, tears, scratches, or fading shall not render the Merchandise therein "defective."

(ii) "On-Order Merchandise" means merchandise currently ordered by Merchant and listed on <u>Exhibit 5.1</u> but which has not been received in the Stores prior to the Sale Commencement Date but which is received at the Stores within 10 days following the Sale Commencement Date. Following such 10th day, On-Order Merchandise (if applicable) arriving at the Stores shall be valued at the Retail Price applicable to such items, less the prevailing Sale discount at the time of delivery to the Stores. Any increase to the Guaranteed Amount relative to On-Order Merchandise (if applicable) delivered to the Stores following the Inventory Date shall be reconciled by the parties as part of the weekly Sale reconciliation. Nothing herein shall obligate the Merchant to purchase or include On-Order Merchandise in the Sale.

(iii) "Excluded Defective Merchandise" means (i) spare parts, component parts, Supplies and other items not intended to be sold to customers as a complete item of merchandise, (ii) returned to vendor inventory located at the Little Rock, AK distribution center, (iii) inventory currently at RMI, the domestic refurbisher in Chicago, IL, (iv) refurbished inventory currently overseas, (v) domestic inventory that has yet to be sent out to be refurbished from the Stores, (vi) inventory already designated as return to vendor or marked "out of stock", and (vii) those items of Defective Merchandise that are not saleable because they are so damaged or defective that such inventory cannot be used for the purpose for which they were intended.

5.2. <u>Valuation</u>.

(a) For purposes of this Agreement, "Retail Price" shall mean the lower of (i) the Merchant's "register" price, or (ii) the ticketed, marked or shelf price of each item of Merchandise, in each case as of the Sale Commencement Date. Notwithstanding the foregoing, the "<u>Retail Price</u>" of items of Defective Merchandise shall be mutually agreed by the Merchant and the Agent.

(b) The Retail Price of any item of Merchandise shall not be reduced further by Sales Taxes. In determining the Retail Price of any Merchandise, the parties shall not rely on, or in any way consider, reductions in the price of Merchandise sold based on employee or similar discounts.

(c) In the event of a conflict between this Agreement and the Inventory Taking Instructions, the terms of this Agreement shall control. If an item is mismarked, the mismarked price shall apply only to that item.

5.3. Excluded Goods. The Merchant shall retain all responsibility for any goods not included as "Merchandise" hereunder (including items of Defective Merchandise for which the Merchant and the Agent cannot agree upon a Retail Price) (collectively, the "Excluded Goods"). Agent shall dispose of the Excluded Goods either by (x) offering such Excluded Goods for sale as "Merchant Consignment Goods" at prices established by the Agent during the Sale, or (y) destroying such Excluded Goods to the extent they are not saleable. It is understood that that the Excluded Goods may not be sold in the Stores and may be sold in bulk to third parties outside the United States and Canada. The Agent shall retain 20% of the sale price plus all expenses, including any expenses associated with the removal, packaging, and delivery of the Merchant Consignment Goods (less Sales Taxes) for all sales of Merchant Consignment Goods,

and Merchant shall receive 80% of the receipts in respect of such sales (less Sales Taxes and all expenses, including any expenses associated with the removal, packaging, and delivery of the Merchant Consignment Goods). Merchant shall receive its share of the receipts of sales of Merchant Consignment Goods on a weekly basis, immediately following the weekly Sale reconciliation by Merchant and Agent pursuant to Section 8.7 below. Except as expressly provided in this Section 5.3, the Agent shall have no cost, expense or responsibility in connection with any Excluded Goods.

5.4. Inventory Taking. The Merchant and the Agent shall use best efforts to cause to be taken a SKU and retail physical inventory of the Merchandise located in the Stores (the "Inventory Taking"), subject to the availability of the Inventory Taking Service (defined below) which Inventory Taking shall be completed in all of the Stores no later than 10 days after the Sale Commencement Date (the "Inventory Completion Date," the date of the Inventory Taking at each Store being the "Inventory Date" for each such Store). The Merchant and the Agent shall jointly employ a mutually acceptable independent inventory taking service (the "Inventory Taking Service") to conduct the Inventory Taking. The Inventory Taking shall be conducted in accordance with the procedures and instructions attached hereto as Exhibit 5.4 (the "Inventory Taking Instructions"). The Merchant and the Agent may each have representatives present during the Inventory Taking, and shall each have the right to review and verify the listing and tabulation of the Inventory Taking Service, provided that senior representatives of the Merchant and the Agent shall be personally present at the Inventory Taking at the first five Stores in order to establish the standards for the Inventory Taking in the remaining Stores. The Agent shall be responsible for 50% of the fees and expenses of the Inventory Taking Service as an Expense hereunder and the Merchant shall be responsible for the remaining fees and expenses

of such Inventory Taking Service. Except as provided in the preceding sentence, the Merchant and the Agent shall each bear their respective costs and expenses relative to the Inventory Taking. The Inventory Taking shall, in any event, be reconciled within seven (7) days after its completion (and the Agent and the Merchant shall use their reasonable best efforts to accomplish such reconciliation within such seven-day period). The Merchant and the Agent agree that during the conduct of the Inventory Taking in each of the Stores, the applicable Stores shall be closed to the public and no sales or other transactions shall be conducted. The Merchant and the Agent agree to cooperate with each other to conduct the Inventory Taking commencing at a time that would minimize the number of hours that such locations would be closed for business.

5.5. <u>Gross Rings</u>. For the period from the Sale Commencement Date until the Inventory Date for each Store, the Agent and the Merchant shall jointly keep (i) a strict count of gross register receipts less applicable Sales Taxes ("<u>Gross Rings</u>") and (ii) cash reports of sales within such Stores. Register receipts shall show for each item sold the Retail Price for such item and the markdown or discount, if any, specifically granted by the Agent in connection with such Sale. All such records and reports shall be made available to the Agent and the Merchant during regular business hours upon reasonable notice. The Agent shall pay that portion of the Guaranteed Amount calculated on the Gross Rings basis, to account for shrinkage, on the basis of 102% of the aggregate Retail Price of Merchandise.

Section 6. <u>Sale Term</u>.

6.1. <u>Term</u>. Subject to satisfaction of the conditions precedent set forth in Section 10 hereof, the Sale shall be deemed to have commenced at each Store on the first day after Bankruptcy Court approval of the Sale but no later than June 1, 2008, (such date with respect to each Store being the "<u>Sale Commencement Date</u>"). The Agent shall complete the Sale

at each Store and vacate such Store in broom-clean condition by no later than August 31, 2008, unless the Sale is extended by mutual written agreement of the Agent and the Merchant (the "<u>Sale Termination Date</u>" and the period from the Sale Commencement Date to the Sale Termination Date as to each Store being the "<u>Sale Term</u>"). The Agent may, in its discretion, terminate the Sale at any Store at any time upon not less than 10 days' prior written notice to the Merchant; provided that the Agent shall not conclude the Sale at any Store prior to June 30, 2008 without Merchant's prior written consent. In the event the Agent fails to provide the Merchant with the requisite notice, the Agent shall be liable for and reimburse the Merchant for applicable Occupancy Expenses for the days by which notice of a Store closing was shortened.

6.2. <u>Vacating the Stores</u>. The Agent shall provide the Merchant with not less than 10 days' written notice of its intention to vacate any Store (as to each such location, the "Vacate Date"). By the applicable Vacate Date, the Agent shall surrender and deliver the Store premises and Store keys to the Merchant or its representatives or assignee and leave such Store in "broom clean" condition, ordinary wear and tear excepted. Subject to the preceding sentence, all assets of the Merchant used by the Agent in the conduct of the Sale (e.g., Supplies, FF&E, etc.) may be abandoned at the Stores at the end of the Sale Term to the extent the same have not been used or sold in the conduct of the Sale. The Agent shall be responsible for all Occupancy Expenses (irrespective of any per diem cap on Occupancy Expenses) for a Store for which the Merchant is or becomes obligated resulting from the Agent's failure to timely vacate such Store. The Agent shall use its reasonable efforts to vacate the Stores no later than 3 days before the end of a calendar month, but in any event, the Agent's obligations to pay all Expenses, including Occupancy Expenses, for each Store shall continue until the later of (a) the applicable Vacate

Date for such Store, or, (b) the 15th day of the calendar month in which the Vacate Date for such Store occurs. On or before the applicable Vacate Date, the Agent shall remove any and all banners and/or signage used by it during the Sale Term. The Agent agrees that it shall be obligated to repair any damage caused by the Agent (or any representative, agent or licensee thereof) to any Store during the Sale Term, including without limitation, damage caused by installation or removal of banners and/or signage and/or removal of FF&E, ordinary wear and tear excepted, and shall indemnify and hold the Merchant and its officers, directors, employees, agents and representatives harmless from and against all claims, demands, penalties, losses, liability or damage, including, without limitation, reasonable attorneys' fees and expenses, asserted directly or indirectly against the Merchant resulting from, or related to (including acts or omissions of persons or entities affiliated with or acting on behalf of the Agent) any such damage or the Agent's breach or failure to leave a Store in "broom clean condition."

Section 7. <u>Sale Proceeds</u>.

7.1. <u>Proceeds</u>. For purposes of this Agreement, "Proceeds" shall mean the aggregate of (i) the total amount (in dollars) of all sales of Merchandise made under this Agreement, exclusive of Sales Taxes; (ii) all proceeds of the Merchant's or the Agent's insurance for loss or damage to Merchandise or loss of cash arising from events occurring during the Sale Term; (iii) the amounts referred to as Proceeds in Section 8.2 hereof, (iv) all proceeds from the sale of FF&E, (v) all proceeds from the sale of Merchandise..

7.2. <u>Deposit of Proceeds (Bank Accounts)</u>. Within twenty-one (21) days after the Sale Commencement Date, Agent may establish its own accounts, dedicated solely for the deposit of the Proceeds and the disbursement of amounts payable to Agent hereunder (the "Agency Accounts") and Merchant shall promptly upon Agent's request execute and deliver all necessary documents to open and maintain the Agency Accounts; provided, however, Agent may elect to continue to use Merchant's Designated Deposit Accounts (as defined below) as the Agency Accounts. The Agency Accounts shall be dedicated solely to the deposit of Proceeds and the disbursement of amounts payable hereunder, and Agent shall exercise sole signatory authority and control with respect to the Agency Accounts. Upon request, Agent shall deliver to Merchant copies of all bank statements and other information relating to such accounts. The Agency Accounts shall be dedicated solely to the deposit of Proceeds and the disbursement of amounts payable hereunder, and Agent shall exercise sole signatory authority and control with respect to the Agency Accounts. Upon Agent's designation of the Agency Accounts, all Proceeds of the Sale (including credit card proceeds) shall be deposited into the Agency Accounts. During the period between the Sale Commencement Date and the date Agent establishes the Agency Accounts, all Proceeds of the Sale (including credit card proceeds), shall be collected by Agent and deposited on a daily basis into depository accounts designated by Merchant for the Stores, which accounts shall be designated solely for the deposit of Proceeds of the Sale (including credit card proceeds), and the disbursement of amounts payable by Agent hereunder (the "Designated Deposit Accounts"). Following the payment of the Initial Guaranty Amount and the posting of the Guaranty L/C and the Expense L/C and on each business day thereafter (or as soon thereafter as is practicable, but in no event less than weekly), Merchant shall promptly pay to Agent by wire funds transfer all collected funds constituting Proceeds deposited into the Designated Deposit Accounts (but not any other funds, including, without limitation, any proceeds of Merchant's inventory sold prior to the Sale Commencement Date).

7.3. <u>Credit Card Proceeds; Other Matters</u>. To the extent available from credit card processors, the Agent shall have the right to use the Merchant's credit card facilities (including the Merchant's credit card terminals and processor(s), credit card processor coding, the Merchant identification number(s), and existing bank accounts) for credit card Proceeds relating solely to the Sale and in conformity with the procedures and practices currently employed by Merchant with respect thereto. All credit card Proceeds relating solely to the Sale shall be paid into the Merchant's existing accounts and disbursed as provided in Section 3.3 hereof.

Section 8. <u>Conduct of the Sale</u>.

8.1. <u>Rights of the Agent</u>. The Agent shall be permitted to conduct the Sale as a "going-out-of-business," "store closing," "bankruptcy," or similarly themed sale throughout the Sale Term. The Agent shall conduct the Sale in the name of and on behalf of the Merchant in a commercially reasonable manner and in compliance with the terms of this Agreement and the Sale Order. The Agent shall provide at the Sale Commencement Date such full time qualified and experienced supervisors engaged by the Agent as independent contractors as Agent deems necessary to oversee the Sale at the Stores, and shall continue to provide qualified and experienced supervisors, subject to reductions for Store closings, through the Sale Termination Date. In addition to any other rights granted to the Agent hereunder in conducting the Sale, but subject to any applicable agreements to which the Merchant is a party (including the leases in respect of the Stores), except as modified by the Sale Order, the Agent, in the exercise of its reasonable discretion, shall have the right:

(a) to establish Sale prices and Stores hours which are consistent with the terms of applicable leases and local laws or regulations, including without limitation Sunday closing laws;

(b) to transfer Merchandise between Stores, provided that, Merchant and Agent shall agree on procedures to record the transfers, receipts and deliveries of Merchandise to and from distributions centers, Stores (excluding distribution centers) and to customers so as to insure that each item of Merchandise is properly accounted for in the Inventory Taking;

(c) except as otherwise expressly included as an Expense, to use without charge during the Sale Term all FF&E, bank accounts (provided that the Merchant and the Agent shall jointly account for all funds pertaining to sales made prior to the Sale Commencement Date), computer hardware and software, existing Supplies located at the Stores, Store keys, case keys, security codes, and safe and lock combinations required to gain access to and operate the Stores, advertising materials, and any other assets of the Merchant (whether owned, leased, or licensed) consistent with applicable terms of leases or licenses (except as modified by the Sale Order). The Agent will deliver to Merchant all keys, codes, and combinations at the end of the Sale. Agent shall be permitted to leave all items of FF&E, materials and Supplies in place in the Stores that have not been expended or sold through the Sale;

(d) to use (i) designated areas of the Merchant's central office facilities, central administrative services, and certain personnel to process payroll, perform MIS and provide other central office services necessary for the Sale, at no additional cost to Agent in

excess of the Expense amount set forth in Section 4.1; and (ii) available offices located at the Merchant's central office facility;

(e) to establish and implement advertising, signage, and promotion programs consistent with the "going-out-of-business," "store closing," "bankruptcy," or similar theme (including, without limitation, by means of media advertising, A-frame, similar interior and exterior signs and banners, and use of sign walkers) in a manner consistent with the Sale Order; and

(f) to supplement the Merchandise at the Stores with Additional Agent Merchandise in accordance with Section 8.9 hereof.

8.2. <u>Terms of Sales to Customers</u>. Although returns of Merchandise are permitted pursuant to Section 8.5, all receipts for sales of Merchandise will state that such sales are "final sales" and "as is." The Agent shall not warrant the Merchandise in any manner, but will, to the extent legally permissible, pass on all manufacturers' warranties to customers. All sales will be made only for cash or with nationally recognized bank credit cards, <u>provided that</u>, if the Agent determines to accept personal checks, the Agent shall bear the risk of loss therefor. If and to the extent requested by the Merchant in writing, the Agent shall accept the Merchant's gift certificates/gift cards, Store credits, due bills, rain checks, Merchandise certificates, Credit Rebate Certificates, and other promotional items or certificates providing the customer with an additional discount on Store Merchandise which have been issued by the Merchant. In the vent that the Bankruptcy Court shall order that the Merchant honor or redeem gift certificates/gift card, store credits, due bills, rain checks, Merchandise certificates, Credit Rebate Certificates, and other promotional items or certificates, Credit Rebate Certificates, and other promotional items or certificates, Credit Rebate Certificates, and other promotional items or certificates, Credit Rebate Certificates, and other promotional items or certificates, Credit Rebate Certificates, and other promotional items or certificates, Credit Rebate Certificates, and other promotional items or certificates (collectively, "Certificates"), the redemption value of such Certificates shall constitute Proceeds for payment of this agreement and shall be reimbursed

by the Merchant to the Agent during the weekly reconciliations provided under Section 8.6 thereof.

8.3. Sales Taxes. During the Sale Term, all sales, excise, gross receipts, and other taxes attributable to sales of Merchandise (other than taxes on income) payable to any taxing authority having jurisdiction (collectively, the "Sales Taxes") shall be added to the sale price of Merchandise and collected by the Agent, on the Merchant's behalf, at the time of sale. Notwithstanding anything in this Agreement or the APA to the contrary, the Merchant shall collect and be entitled to maintain any sales tax disbursement fees. All Sales Taxes shall be deposited as instructed by the Merchant. Provided that the Agent has collected all Sales Taxes during the Sale and remitted the proceeds thereof to the Merchant, the Merchant shall prepare and file all applicable reports and documents required by the applicable taxing authorities and the Merchant shall promptly pay when due all Sales Taxes. The Merchant and the Agent will agree to the computation of gross receipts for verification of all such tax collections. Provided that the Agent performs its responsibilities in accordance with this Section 8.3, the Merchant shall indemnify and hold harmless the Agent, and its officers, directors, shareholders and agents from and against any and all costs, including, but not limited to, reasonable attorneys' fees, assessments, fines, or penalties which the Agent sustains or incurs as a result or consequence of the failure by the Merchant to promptly pay when due such taxes to the proper taxing authorities and/or the failure by the Merchant to promptly file when due with such taxing authorities all reports and other documents required, by applicable law, to be filed with or delivered to such taxing authorities. If the Agent fails to perform its responsibilities in accordance with this Section 8.3, and provided the Merchant complies with its obligations hereunder, the Agent shall indemnify and hold harmless Merchant, and its officers, directors, and agents from and against

any and all costs including, but not limited to, reasonable attorneys' fees, assessments, fines, or penalties which the Merchant sustains or incurs as a result or consequence of the failure by the Agent to collect Sales Taxes and/or, the failure by the Agent to promptly deliver any and all reports and other documents required to enable the Merchant to file any requisite returns with such taxing authorities.

8.4. <u>Supplies</u>. The Agent shall have the right to use, without charge, all existing supplies located at the Stores, including, without limitation, boxes, bags, paper, twine, and similar sales materials (collectively, the "Supplies"). In the event that additional Supplies are required in any of the Stores during the Sale, the Merchant agrees to promptly provide the same, if available, to the Agent which shall constitute an Expense of the Sale. The Merchant does not warrant that the existing Supplies as of the Sale Commencement Date are adequate for the purposes of the Sale. All unused Supplies shall remain in the Stores at the Sale Termination Date.

8.5. <u>Returns of Merchandise</u>. The Agent shall accept returns of Merchandise sold by Merchant prior to the Sale Commencement Date if requested to do so by the Merchant; <u>provided that</u> (i) the customer has the original register receipt, and (ii) such return complies with the Merchant's return policy existing immediately prior to the Sale Commencement Date, which are provided to the Agent in writing on or prior to the Sale Commencement Date. The Agent shall treat these returns in a manner consistent with the Merchant's ordinary course procedures. If such returned Merchandise is saleable as first-quality Merchandise, it shall be included in the Merchandise and valued at the Retail Price applicable to such item less the prevailing Sale discount at the time of the return (and, in the event that there is no similar Merchandise already located in the Store, then the prevailing discount applicable to such returned Merchandise shall be an average of the discount applicable to all Merchandise in the Store at the time of receipt of the returned Merchandise), and the Guaranteed Amount shall be adjusted accordingly. Refunds for cash shall be reimbursed by the Merchant to the Agent during the weekly reconciliations provided under Section 8.6.

8.6. <u>Certain Procedures</u>. As soon as practicable, the Agent and the Merchant shall each designate responsible parties with respect to each of the following areas: (i) financial operations, (ii) store operations, (iii) information technology, (iv) human resources, and (v) executive designees. All communications with respect to these designated areas shall be copied to the parties designated by the Agent and the Merchant as responsible parties for such areas. On each Monday during the Sale Term, commencing on the second Monday after the Sale Commencement Date, the Agent and the Merchant shall cooperate to reconcile, with respect to the previous week, Expenses of the Sale, and such other Sale-related items as either party shall reasonably request, including, but not limited to, establishing procedures for closing and surrendering the Stores, in each case for the prior week or partial week (i.e., Thursday through Wednesday), all pursuant to procedures mutually agreed upon by the Merchant and the Agent.

8.7. <u>Force Majeure</u>. If any act of God or other casualty outside of the Agent's reasonable control, specifically excluding, however, any such events to the extent arising out of (i) the Agent's acts or omissions, or (ii) routine market or weather-related conditions (outside of earthquakes and extraordinary meteorological and other extraordinary events), prevents the conduct of business in the ordinary course at any Store for a period in excess of five consecutive business days, such Store and the Merchandise located at such Store shall be eliminated from the Sale and considered to be deleted from this Agreement and the APA as of the last date of such event (unless the Agent and the Merchant otherwise agree in writing), and the Agent and the

Merchant shall have no further rights or obligations hereunder with respect thereto except as to reconciling the Expenses of the Sale at such Store and such other Sale-related items as either party shall reasonably request; <u>provided</u>, <u>however</u>, <u>that</u> the proceeds of any insurance attributable to such Merchandise shall constitute Proceeds hereunder. The Agent will use its best efforts to consolidate and transfer all Merchandise which is not the subject of insurance proceeds and include said Merchandise in the Sale in other Stores.

8.8. <u>Merchant's Right of Access</u>. The Merchant shall have the right to be present in the Stores at any time.

Section 9. <u>Employee Matters</u>.

9.1. <u>Merchant's Employees</u>. Subject to the terms of any employment contract, and consistent with the Merchant's past practices, policies and procedures relating to the employment of its employees, in each case, which is provided to the Agent in writing or otherwise made available to the Agent, the Agent may use the Merchant's employees in the conduct of the Sale to the extent the Agent deems expedient and the Agent may select and schedule the number and type of the Merchant's employees required for the Sale. The Merchant shall use reasonable efforts to cause the Merchant's employees to cooperate with the Agent and the Agent's supervisors. The Agent shall identify any such employees to be used in connection with the Sale (each such employee, a "Retained Employee") on or prior to the Sale Commencement Date. Employees will be selected by seniority and status where possible. Agent acknowledges and agrees that the selection and scheduling of Retained Employees and the decision to cease using Retained Employees in connection with the Sale shall be made with due regard to the Merchant's desire to minimize severance, termination and any WARN Act costs to the Merchant, and shall be made so as not to interrupt any statutory working notice.

Notwithstanding the foregoing, the Merchant's employees shall at all times remain employees of the Merchant; provided, however, it is understood and agreed that the Agent's on-site supervisors shall not be employees of Merchant under any circumstances. The Agent's selection and scheduling of the Merchant's employees shall at all times comply with all applicable laws and regulations, including Merchant's collective bargaining agreements, if any. The Merchant and the Agent agree that except to the extent that wages and benefits of Retained Employees constitute Expenses hereunder, nothing contained in this Agreement and none of the Agent's actions taken in respect of the Sale shall be deemed to constitute an assumption by the Agent of any of the Merchant's obligations relating to any of Merchant's employees including, without limitation, Excluded Benefits, WARN Act claims, and other termination type claims and obligations, or any other amounts required to be paid by statute or law; nor shall Agent become liable under any collective bargaining or employment agreement or be deemed a joint or successor employer with respect to such employees. The Agent shall comply in the conduct of the Sale with all of the Merchant's employee rules, regulations, guidelines, and policies which have been provided to Agent in writing on or prior to the Sale Commencement Date. Except as disclosed to the Agent in writing prior to the execution of this Agreement by the parties hereto, the Merchant shall not, without the prior consent of Agent, raise the salary or wages or increase the benefits for, or pay any bonuses or other extraordinary payments to, any of the Store employees prior to the Sale Termination Date. The Merchant may transfer any Retained Employee during the Sale Term upon ten (10) days' notice to the Agent and with the Agent's prior consent, which consent shall not be unreasonably withheld. Notwithstanding the foregoing, the Merchant shall be entitled to transmit notices required under the WARN Act to its employees

in its sole discretion. Nothing in this Agreement or the APA shall alter the employees' at-will employment relationship with the Merchant.

9.2. Termination of Employees. Consistent with the Merchant's past practices, policies and procedures related to the employment of its employees, the Agent may in its discretion stop using any Retained Employee, subject to the conditions provided for herein. In the event that the Agent desires to cease using any Retained Employee, the Agent shall notify Merchant at least seven days prior thereto so that the Merchant may coordinate the termination of such employee; provided that, in the event that the Agent determines to cease using an employee "for a failure to satisfactorily perform his/her duties" (examples of which are, in summary, dishonesty, fraud, or breach of employee duties), the seven day notice period shall not apply, provided, further, that the Agent shall immediately notify the Merchant of the basis for such "cause" so that the Merchant can arrange for termination of such employee. From and after the date of this Agreement and until the Sale Termination Date, the Merchant shall not transfer or dismiss employees of the Stores except "for a failure to satisfactorily perform his/her duties" without Agent's prior consent. Notwithstanding the foregoing, the Agent (i) shall not have the right to terminate the actual employment of any employee, but rather may only cease using such employee in the Sale and paying any Expenses with respect to such employee after notice to the Merchant, and (ii) will indemnify and hold the Merchant harmless with respect to any claims by Retained Employees arising from the Agent's treatment of such Retained Employees.

9.3. <u>Payroll Matters</u>. During the Sale Term, the Merchant shall process the base payroll for all Retained Employees as well as any former employees and temporary labor retained by Agent.

9.4. Employee Retention/Performance Bonuses. The Agent and the Merchant shall determine, as an Expense of the Sale, retention/performance bonuses ("Performance Bonuses") (which bonuses shall be inclusive of payroll taxes but as to which no benefits shall be payable), of up to ten percent (10%) of base payroll, to be paid to certain Retained Employees who do not voluntarily leave employment and are not terminated "for failure to satisfactorily perform their duties." The amount of such Performance Bonuses shall be in an amount to be determined by the Agent and Merchant pursuant to an incentive plan mutually agreeable to the Merchant and the Agent. Performance Bonuses shall be payable by Merchant within 30 days after the Sale Termination Date and shall be processed through Merchant's payroll system. The Agent shall provide the Merchant with a copy of the Agent's Performance Bonus plan for both the Merchant's employees and the Agent's supervisors no later than the Sale Commencement Date, in order to permit Merchant to coordinate such plan with its existing severance and/or retention plans for its employees.

Section 10. Conditions Precedent to Closing.

10.1. <u>Conditions Precedent</u>. The willingness of the Agent and the Merchant to enter into the transactions contemplated under this Agreement and the APA are directly conditioned upon the satisfaction of the conditions set forth in the APA and the following conditions at the time or during the time periods indicated, unless specifically waived in writing by the applicable party:

(a) All representations and warranties of the Merchant and the Agent hereunder and in the APA shall be true and correct in all material respects as of the date hereof and as of the Sale Commencement Date.

(b) The Bankruptcy Court shall have entered the Sale Order on or before June 2, 2008.

(c) The Agent and the Merchant shall have received from the other (in form and substance reasonably satisfactory to the Agent and the Merchant) such documents or instruments as each party may reasonably request to fully effect the transfer of the Designation Rights and to otherwise consummate the transactions contemplated by this Agreement and the APA.

Section 11. <u>Representations, Warranties, and Covenants.</u>

11.1. <u>Merchant's Representations, Warranties, and Covenants</u>. The Merchant hereby represents, warrants, and covenants in favor of Agent (in addition to those contained in the APA) as follows:

(a) The Merchant is, and during the Sale Term will continue to be, duly authorized and qualified to do business in each jurisdiction where the nature of its business or properties requires such qualification, including all jurisdictions in which the Stores are located, except, in each case, to the extent that the failure to be in good standing or so qualified could not reasonably be expected to have a material adverse effect on the ability of the Merchant to execute and deliver this Agreement and the APA and perform fully its obligations and/or conduct of the Sale.

(b) Except as may be required in connection with the issuance of any Sale Order: (i) the Merchant has the right, power, and authority to execute and deliver this Agreement and each other document and agreement contemplated hereby, including, without limitation, the APA (collectively, together with this Agreement, the "Sale Documents") and to perform fully its obligations thereunder; (ii) the Merchant has taken all necessary actions

required to authorize the execution, delivery, and performance of the Sale Documents, and no further consent or approval is required for the Merchant to enter into and deliver the Sale Documents, to perform its obligations thereunder, and to consummate the Sale, except for any such consent the failure of which to be obtained could not reasonably be expected to have a material adverse effect on the ability of the Merchant to execute and deliver this Agreement and perform fully its obligations hereunder and/or conduct of the Sale; and (iii) each of the Sale Documents has been duly executed and delivered by the Merchant and constitutes the legal, valid and binding obligation of Merchant enforceable in accordance with its terms.

(c) The Merchant owns and will own at all times during the Sale Term, good and marketable title to all of the Merchandise and FF&E (such FF&E being identified in Exhibit 12.1(c)(1)) free and clear of all liens, claims, and encumbrances of any nature, other than the liens listed in Exhibit 12.1 and any applicable statutory liens. The Merchant shall not create, incur, assume, or suffer to exist any security interest, lien, or other charge or encumbrance upon or with respect to any of the Merchandise, the FF&E or the Proceeds other than as provided for herein (including those listed in Exhibit 12.1). The Sale Order shall provide that (following payment of the Initial Guarantee Payment and delivery of the Guarantee L/C and the Expense L/C) all such liens, claims, and encumbrances shall be transferred to and attach only to the Guarantee Amount, Expenses, and any other amounts due to the Merchant hereunder.

(d) (i) The Merchant has maintained its pricing files in the ordinary course of business, and prices charged to the public for goods, including the "register price," are the same in all material respects as set forth in such pricing files for the periods indicated therein (including Merchant's catalogue and website, but excluding any "circular" or specially advertised Merchandise); (ii) there are no material changes to the pricing files and records posted to the on-line dataroom, and no material changes to prices other than those reflected in the pricing files and records have been made; and (iii) all such pricing files and records are true and accurate in all material respects as to the selling price to the public for such goods (inclusive of prices in the Merchant's catalogue and website, but excluding specially advertised Merchandise and without consideration of any point of sale markdowns) as of the dates and for the periods indicated therein.

(e) All normal course permanent markdowns (price changes) on the Merchandise located at the Stores and Transferred Merchandise have been taken and have been programmed into Merchant's registers and point of sales equipment.

(f) The Merchant has not in contemplation of the Sale marked up or raised the price of any items of Merchandise, or removed or altered any tickets or any indicia of clearance merchandise, in each case in contemplation of the Sale. The "register" price represents the selling price of each item of Merchandise reflected in the pricing files posted to the on-line dataroom, taking into account all "point of sale" markdowns and promotions and the status of any applicable item of Merchandise as "refurbished." As of the Sale Commencement Date, the sale prices of Merchandise located in the Stores is consistent in all material respects as the selling prices for such Merchandise in the Merchant's ongoing store locations.

(g) Since May 1, 2008, the Merchant has not and shall not purchase or transfer to or from the Stores any Merchandise or goods outside the ordinary course, except as contemplated by this Agreement and as otherwise set forth herein and except to the extent that such matters have been impacted by the commencement of the Bankruptcy Case.

(h) To the best of the Merchant's knowledge, all Merchandise is in compliance with all applicable federal, state, or local product safety laws, rules, and standards. The Merchant shall provide the Agent with its historic policies and practices, if any, regarding product recalls prior to the Sale Commencement Date and shall notify Agent of any product recalls it becomes aware of after the Sale Commencement Date.

(i) Throughout the Sale Term, and after giving effect to the Sale Order, the Agent shall have the right, subject to the applicable agreements to which the Merchant is a party (except as provided in the Sale Order), to the unencumbered use and occupancy of, and peaceful and quiet possession of, each of the Stores, the assets currently located at the Stores, and the utilities and other services provided at the Stores. The Merchant shall throughout the Sale Term maintain in good working order, condition, and repair, all cash registers, heating systems, air conditioning systems, elevators, escalators, and all other mechanical devices necessary for the conduct of the Sale at the Stores. Except any amounts owing as a result of the commencement of the Bankruptcy Case, and absent a bona fide dispute, throughout the Sale Term, as long as the Agent reimburses the Merchant for Expenses as provided herein, the Merchant shall remain current on all expenses and payables necessary for the conduct of the Sale (other than those relating to any period prior to the commencement of the Bankruptcy Case).

(j) Except any amounts owing as a result of the commencement of the Bankruptcy Case, the Merchant has paid, and as long as the Agent reimburses the Merchant for Expenses as provided herein, the Merchant will continue to pay throughout the Sale Term, all self-insured or the Merchant-funded employee benefit programs for current Store employees, including health and medical benefits and insurance and all proper claims made or to be made in

accordance with such programs (other than those relating to any period prior to the commencement of the Bankruptcy Case).

11.2. <u>Agent's Representations, Warranties, and Covenants</u>. Each entity comprising the Agent hereby severally represents, warrants, and covenants in favor of Merchant as follows:

(a) The Agent shall conduct the Sale in accordance with applicable non-bankruptcy laws and Merchant's Leases and shall not change the Merchant's procedures and practices currently employed, except as otherwise provided in the Sale Order.

(b) Agent has the right, power, and authority to execute and deliver each of the Agency Documents to which it is a party and to perform fully its obligations thereunder. Agent has taken all necessary actions required to authorize the execution, delivery, and performance of the Agency Documents, and no further consent or approval is required on the part of Agent for Agent to enter into and deliver the Agency Documents, to perform its obligations thereunder, and to consummate the Sale. Each of the Agency Documents has been duly executed and delivered by the Agent and, constitutes the legal, valid, and binding obligation of Agent enforceable in accordance with its terms. No court order or decree of any federal, state, or local governmental authority or regulatory body is in effect that would prevent or impair or is required for Agent's consummation of the transactions contemplated by this Agreement, and no consent of any third party which has not been obtained is required therefor other than as provided herein. No contract or other agreement to which Agent is a party or by which Agent is otherwise bound will prevent or impair the consummation of the transactions contemplated by this Agreement.

(c) The Agent shall provide qualified and experienced store supervisors to conduct the Sale.

(d) Prior to the execution of this Agreement and the APA, the Merchant has provided the Agent reasonable access to all pricing and cost files, computer hardware, software and data files, inter-Store transfer logs, markdown schedules, invoices, style runs and all other documents relative to the price, mix and quantities of inventory located at the Stores which have been posted by the Merchant to the on-line data room made available to prospective bidders for the right to conduct the Sale.

(e) On the date immediately preceding the Inventory Date, the Agent has had and will have had the opportunity to inspect the Stores and the Merchandise.

- 11.3. <u>Additional Covenants</u>.
 - (a) <u>Confidentiality</u>.

(i) Except as modified by the APA, All information concerning the Agent received by the Merchant (other than that information that is a matter of public knowledge or that has been published for public distribution or filed as public information with any governmental authority) shall be used by the Merchant only to evaluate and obtain approval of this transaction and shall not at any time, except in filings with the Bankruptcy Court in connection with this Agreement and the transactions contemplated hereby, be used for the advantage of or disclosed by the Merchant to any third person without the prior written consent of the Agent unless otherwise required by law (in which event, the Merchant shall notify the Agent prior to any such disclosure). The Merchant may disclose such information on a confidential basis to employees, officers, agents, auditors, consultants, coursel to the Merchant, directors of the Merchant and the Committee and its professionals and any secured lender to the Merchant and their professionals.

(ii) All information concerning the Merchant and their assets received by the Agent (other than that information that is a matter of public knowledge or that has been published for public distribution or filed as public information with any governmental authority) shall be used by the Agent only to evaluate and obtain approval of this transaction and shall not at any time be used for the advantage of or disclosed by the Agent to any third person without the prior written consent of the Merchant unless otherwise required by law (in which event, the Agent shall notify the Merchant prior to any such disclosure). The Agent may disclose such information on a confidential basis to employees, officers, agents, auditors, consultants, counsel and directors of the Agent.

(iii) The covenants set forth in this Section 12.3(a) shall survive the Closing or termination of this Agreement and the APA.

(b) <u>Transaction Expenses</u>. Except as expressly provided for herein, each party shall pay its own fees, costs and expenses incurred with respect to this Agreement and the APA, whether or not the transactions contemplated hereby are consummated.

(c) <u>Further Assurances</u>. The Agent and the Merchant shall, from time to time after the Closing, without further consideration, execute and deliver such instruments and take such further actions as may be reasonably necessary or desirable to carry out the provisions hereof and the transactions contemplated hereby.

Section 12. <u>Insurance</u>.

12.1. <u>Merchant's Liability Insurance</u>. The Merchant shall continue until the Sale Termination Date, in such amounts as it currently has in effect, all of its liability insurance

policies including, but not limited to, products liability, commercial general liability, auto liability, and umbrella liability insurance, covering injuries to persons and property in, or in connection with the Merchant's operation of the Stores, and shall cause the Agent to be named as an additional insured (as its interest may appear) with respect to all such policies concerning the Merchant's operation of the Stores. The Merchant shall deliver to the Agent certificates evidencing such insurance setting forth the duration thereof and naming the Agent as an additional insured, in form reasonably satisfactory to Agent. All such policies shall require at least 30 days' prior notice to the Agent of cancellation, non-renewal, or material change during the Sale Term.

12.2. <u>Merchant's Casualty Insurance</u>. The Merchant will provide throughout the Sale Term, at the Agent's cost as an Expense hereunder (limited to the amount set forth in Section 4.1(q) above), fire, flood, theft, and extended coverage casualty insurance covering the Merchandise in a total amount at any time equal to no less than the aggregate Retail Price of the Merchandise at such time, subject to any self-insurance amounts and the amount of any deductible or self-insured retention, which shall be paid by the Merchant to the Agent in the event of any casualty loss affecting the Merchandise. From and after the date of this Agreement and the APA until the Sale Termination Date, all such policies will also name the Agent as loss payee (as its interest may appear). In the event of a loss to the Merchandise on or after the date of this Agreement, the proceeds of such insurance attributable to the Merchandise plus any selfinsurance amounts and the amount of any deductible or self-insured retention actually paid by the Merchant to the Agent (which amounts shall not be an Expense), shall constitute Proceeds hereunder. If requested by the Agent, the Merchant shall deliver to the Agent certificates evidencing such insurance setting forth the duration thereof and naming the Agent as loss payee (as its interest may appear), in form and substance reasonably satisfactory to the Agent. All such policies shall require at least 30 days' prior notice to the Agent of cancellation, non-renewal, or material change during the Sale Term. The Merchant shall not make any change in the amount of any deductibles or self-insurance amounts prior to the Sale Termination Date without the Agent's prior written consent.

12.3. Agent's Insurance. The Agent shall maintain at the Agent's cost and expense throughout the Sale Term in such amounts as it currently has in effect, comprehensive public liability and automobile liability insurance policies with insurance coverage in an amount reasonably satisfactory to the Merchant covering injuries to persons and property in or in connection with the Agent's agency at the Stores, and shall cause the Merchant to be named as an additional insured with respect to such policies. Exhibit 12.3 lists the Agent's existing policies and the coverage amounts for the policies. If requested by the Merchant, the Agent shall deliver to the certificates evidencing such insurance policies setting forth the duration thereof and naming the Merchant as additional insured, in form and substance reasonably satisfactory to the Merchant. In the event of a claim under any such policies, the Agent shall be responsible for the payment of all deductibles, retentions, or self-insured amounts thereunder, unless it is determined that liability arose by reason of the wrongful acts or omissions or gross negligence of the Merchant or the Merchant's independent contractors or agents, other than the Agent or the Agent's employees, agents or independent contractors. All such policies shall require at least 30 days' prior notice to the Agent of cancellation, non-renewal, or material change during the Sale Term.

12.4. <u>Worker's Compensation Insurance</u>. The Merchant shall at all times during the Sale Term maintain in full force and effect worker's compensation insurance (including

employer liability insurance) covering all Retained Employees in compliance with all statutory requirements.

Section 13. <u>Indemnification</u>.

13.1. Merchant Indemnification. The Merchant shall indemnify and hold the Agent and its officers, directors, employees, agents, and independent contractors harmless from and against all claims, demands, penalties, losses, liability, or damage, including, without limitation, reasonable attorneys' fees and expenses, directly or indirectly asserted against, resulting from, or related to the Merchant's material breach of or failure to comply with any of its agreements, covenants, representations, or warranties contained in any Sale Document. The Merchant shall indemnify the Agent from and against all claims, demands, penalties, losses, liability, or damage, including, without limitation, reasonable attorneys' fees and expenses, directly or indirectly asserted against, resulting from or related to: (i) any structural failures of the buildings; (ii) any harassment or any other unlawful, tortious, or otherwise actionable treatment of any customers, employees or agents of the Agent by the Merchant or any of its representatives; (iii) the gross negligence (including omissions) or willful misconduct of the Merchant or any officer, directors, employees, agents or representative of the Merchant; and (iv) any claims arising from breach of warranty, product liability or product recall relating to any item of Merchandise sold through the Sale.

13.2. <u>Agent Indemnification</u>. The Agent shall indemnify and hold the Merchant and its officers, directors, employees, agents, and representatives harmless from and against all claims, demands, penalties, losses, liability, or damage, including, without limitation, reasonable attorneys' fees and expenses, directly or indirectly asserted against, resulting from, or related to:
(i) Agent's material breach of or failure to comply with any of its agreements, covenants,

representations, or warranties contained in any Sale Document; (ii) any harassment or any other unlawful, tortious, or otherwise actionable treatment of any customers, employees, or agents of the Merchant by Agent or any of its representatives; (iii) any claims by any party engaged by Agent as an employee or independent contractor arising out of such employment; and (iv) the gross negligence (including omissions) or willful misconduct of Agent or officer, directors, employees agents, or representative.

Section 14. <u>Defaults</u>. [Omitted]

Section 15. <u>Miscellaneous</u>.

15.1. <u>Notices</u>. All notices and communications provided for pursuant to this Agreement shall be in accordance with the APA

15.2. <u>Governing Law</u>. This Agreement shall be governed and construed in accordance with the laws governing the APA.

15.3. <u>Amendments</u>. This Agreement may not be modified except in accordance with the APA.

15.4. <u>Successors and Assigns</u>. Section 13.8 of the APA shall apply to this Agreement, *provided*, *however*, that Agent may not assign its rights or obligations relating to the Sale without the prior written consent of Merchant.

15.5. <u>Sale of FF&E</u>. The Agent shall have the right to sell and dispose of all of the FF&E located in the Stores and the corporate headquarters located in San Francisco, CA, and in all distribution centers, and all revenues and proceeds from such sales, if any, shall be for the account of the Agent. The Agent shall bear all expenses of the sale of the FF&E and shall not constitute Expenses hereunder. At the conclusion of the Sale, the Agent may leave in place at the Stores (and in the headquarters and in any distribution center) any unsold items of FF&E.

15.6. Lien. Without limiting the Agent's offset rights under Section 3.3(g) above, upon payment of the Initial Guaranty Payment to the Merchant and the issuance of the Guaranty L/C and the Expense L/C and in consideration of the Agent's payment of the Guaranteed Amount, and Expenses, and the provision of services hereunder to the Merchant, the Merchant hereby grants to the Agent pursuant to Bankruptcy Code § 364(d) a valid security interest in and lien upon the Merchandise, FF&E and the Proceeds to secure all obligations of the Merchant to the Agent hereunder. Until the payment in full of all amounts due to the Merchant hereunder, including, but not limited to, the Guaranteed Amount and reimbursement of Expenses (to the extent owed after consideration of Agent's offset rights under section 3.3(g) hereof, if any), the security interest granted to Agent hereunder shall remain junior to the security interest of the Lender, only to the extent of such unpaid amount (to the extent owed after consideration of Agent's offset rights under section 3.3(g) hereof, if any). Upon entry of the Sale Order and payment of the Initial Guaranteed Payment and the issuance of the Guaranty L/C and the Expense L/C, the security interest granted to Agent hereunder shall be deemed properly perfected without the need for further filings or documentation or Bankruptcy Court order.

15.7. <u>Section Headings</u> The headings of sections of this Agreement are inserted for convenience only and shall not be considered for the purpose of determining the meaning or legal effect of any provisions hereof.

IN WITNESS WHEREOF, the Agent and the Merchant hereby execute this Agreement by their duly authorized representatives as a sealed instrument as of the day and year first written above.

SHARPER IMAGE CORPORATION

By: /s/ Rebecca L. Roedell Name: Rebecca L. Roedell Title: Executive Vice President & Chief Financial Officer

HILCO MERCHANT RESOURCES, LLC,

By: <u>/s/ Eric W. Kaup</u> Name: Eric W. Kaup Title: Executive Vice President, General Counsel, Member

HILCO CONSUMER CAPITAL, LLC,

By: <u>/s/ Eric W. Kaup</u> Name: Eric W. Kaup Title: Executive Vice President, General Counsel, Managing Member

GORDON BROTHERS RETAIL PARTNERS, LLC,

on behalf of Purchaser

By: <u>/s/ Rafael Klotz</u> Name: Rafael Klotz Title: Managing Director

GB BRANDS, LLC,

on behalf of Purchaser

By: /s/ Rafael Klotz

Name: Rafael Klotz Title: Managing Director